

Article



Cooperative Visual-SLAM System for UAV-Based Target Tracking in GPS-Denied Environments: A Target-Centric Approach

Juan-Carlos Trujillo ¹, Rodrigo Munguia ^{1,*}, Sarquis Urzua ¹ and Antoni Grau ²

- ¹ Department of Computer Science, CUCEI, University of Guadalajara, Guadalajara 44430, Mexico; juancarlos_max@hotmail.com (J.-C.T.); isi.sarquis@gmail.com (S.U.)
- ² Department of Automatic Control, Technical University of Catalonia UPC, 08034 Barcelona, Spain; antoni.grau@upc.edu
- * Correspondence: rodrigo.munguia@academicos.udg.mx

Received: 31 March 2020; Accepted: 8 May 2020; Published: 15 May 2020



Abstract: Autonomous tracking of dynamic targets by the use of Unmanned Aerial Vehicles (UAVs) is a challenging problem that has practical applications in many scenarios. In this context, a fundamental aspect that must be addressed has to do with the position estimation of aerial robots and a target to control the flight formation. For non-cooperative targets, their position must be estimated using the on-board sensors. Moreover, for estimating the position of UAVs, global position information may not always be available (GPS-denied environments). This work presents a cooperative visual-based SLAM (Simultaneous Localization and Mapping) system that allows a team of aerial robots to autonomously follow a non-cooperative target moving freely in a GPS-denied environment. One of the contributions of this work is to propose and investigate the use of a target-centric SLAM configuration to solve the estimation problem that differs from the well-known World-centric and Robot-centric SLAM configurations. In this sense, the proposed approach is supported by theoretical results obtained from an extensive nonlinear observability analysis. Additionally, a control system is proposed for maintaining a stable UAV flight formation with respect to the target as well. In this case, the stability of control laws is proved using the Lyapunov theory. Employing an extensive set of computer simulations, the proposed system demonstrated potentially to outperform other related approaches.

Keywords: unmanned aerial vehicles; cooperative visual SLAM; state estimation; target tracking; observability; flight formation control

1. Introduction

Nowadays, Unmanned Aerial Vehicles (UAVs), computer vision techniques, and flight control systems have received great attention from the research community in robotics. This interest has as a consequence the development of systems with a high degree of autonomy. UAVs are very versatile platforms. In particular, aerial robots with rotary wings, as the quadcopters, allow great flexibility of movements, which makes them very useful for several tasks and applications [1,2]. Multi-robot systems have also received great attention from the robotics research community. This attention is motivated by the inherent versatility that those systems have to perform tasks that could present difficulties to be realized by a single robot. The use of several robots can have advantages such as an increase of robustness, better performance, and efficiency [3,4].

In this context, one important research problem is the control and coordination of a team of UAVs flying in formation with respect to a non-cooperative moving target. In this context, a fundamental task that must be addressed to control the flight formation is the estimation of the positions of UAVs and

the moving target. For most applications, GPS (Global Positioning System) still represents the main alternative for addressing the localization problem of UAVs. Nevertheless, the use of GPS presents some drawbacks, for instance, in scenarios where GPS signals are jammed intentionally [5] or when the precision error is substantial and they provide poor operability due to multipath propagation (natural and urban canyons [6,7]). In addition, there exist scenarios (e.g., indoor) where GPS is completely unavailable. Moreover, for non-cooperative targets, their position has to be estimated using on-board sensors.

In such scenarios, when considering a team of aerial robots flying in formation with respect to a particular target, the absolute pose (e.g., world-centric configuration) is not so important compared to the relative position information between the aerial vehicles and the target (e.g., Robot-centric configuration). In this case, sensors can help to estimate the required position information such as range sensors (laser or sonar) and radio-frequency (RF) tag-like-sensors (see [8–11]). However, these kinds of sensors are expensive and sometimes heavy, and their use in outdoor environments is limited somehow. Moreover, some of these sensors need the cooperation of the moving target. Active laser systems (e.g., LiDAR [12]) represent a very interesting sensing technology; they can operate under any visibility condition (i.e., both day and night, unlike cameras) and can directly provide 3D measurements about the surrounding environment. On the other hand, LiDAR is generally expensive overweighting the system for certain applications presenting moving parts which can induce some errors. Stereo systems [13] and depth cameras [14] also can obtain 3D information about the target and the environment; however, this kind of system requires that the objects to be measured are near the sensor, thus considerably limiting their range of application.

In the above context, more works have appeared focusing on the use of cameras to develop navigation systems based on visual information that can operate in periods or circumstances where the GPS is unavailable. Cameras are well suited for their use in embedded systems, and also can be used for estimating the target relative position.

1.1. Related Work

In this work, the use of a cooperative visual-based SLAM scheme is proposed, for addressing the problem of estimating the relative position of the aerial robots with respect to one target, to control the flight formation with respect to the same target.

In the literature, different approaches can be found that use visual information to carry out the control of UAVs (Visual-Servoing): [15–19]. In addition, Refs. [20–23] are examples of approaches where the position of a target is estimated in a probabilistic manner by means of the fusion of monocular measurements from two or more cameras. These works are only focused on the estimation of the target position and assume that there exists an ideal control scheme that lead the robots towards the moving target. In [24–28], different control schemes are presented for addressing the problem of maintaining a desired flight formation with respect to a moving target. A compilation of different techniques for addressing the flight formation control problem is presented in [29]. One of the underlying assumptions in these works is that the position of vehicles is available, either globally referenced or locally referenced.

In [30–32], methods for the tracking, control, and estimation of a target using a UAV with a monocular camera on board are presented. In these works, it is assumed that the geometry of the target is known or its movement is restricted to a plane. In [33,34], tracking control schemes using multi-robot and single-robot configurations are presented. These works make the strong assumption that the relative distances of the vehicles with respect to the target are known. In [35], a scheme for tracking control and estimation of a target using a small UAVs with monocular cameras on-board is presented. In this work, it is assumed that there are available measurements of the altitude of the vehicles with respect to the target. Another single-robot scheme is presented in [36]. This work relies on stereo-vision; however, as it will be shown later, the use of this sensor is unreliable in the present problem that authors state. In [37], a multi-robot system based on an LiDAR is presented.

In [38], a method to control an space robot that follows an uncooperative target using a vision system is presented. The estimation of the position of the target is carried out using an adaptive unscented Kalman filter based on measurements obtained from an homography. In this case, to obtain the homography, the coplanar visual landmarks need to be extracted from the target.

In [39], a single-robot SLAM scheme, where the state of the dynamic target is included in the system state, is presented. In this case, the position of the robot, the map, and the target are estimated using a constrained local submap filter (CLSF) based on an EKF (Extended Kalman Filter) configuration. In [40], the problem of cooperative localization and target tracking with a team of moving robots is addressed. The problem is modeled as a least-squares minimization problem and solved using sparse optimization. Static known landmarks are used to define a common reference frame for the robots and the targets. However, in many applications, it is complicated or even impossible to have prior knowledge about the position of landmarks. In [41], an approach is presented for estimating the position of objects tracked by a team of mobile robots and the use of such objects for a better self-localization. In this case, relationship among objects is used to estimate objects' position given a map. However, the geometric knowledge of objects is required in order to carry out the location of the robots. In [42], a control scheme for commanding the formation with respect to a target is presented. In this case, the local sensor information is used to estimate the relative position and orientation of the robots with respect to the target (target-centric configuration).

1.2. Objectives and Contributions

Recently, some works, such as [43], present the development of observers and controllers for relative estimation and circumnavigation of a target using bearing-only measurements or range with bearing measurements. In that work, the movement of the target is restricted to a plane. In addition, one of the assumptions is that range measurements with respect to the target are available, which is very difficult to obtain for non-cooperative targets. In [44], a Gaussian sum FIR filter (GSFF) to estimate the position of a target is presented. In this work, both estimation and tracking of the target are carried out only in a two-dimensional space. In [45], a UAV-based target tracking and its recognition system are presented. In this work, a geographic information system (GIS) is used to provide geo-location, environmental, and contextual information. The work in [46] presents a distributed control strategy applied to a team of agents to create a prescribed formation with its centroid tracking a moving target. In this work, one of the assumptions is that some relative position measurements between agents and the target are available.

To try avoiding the restrictions given in the previous approaches, in this present work, the use of a cooperative visual-based SLAM scheme is studied for addressing the problem of estimating the relative position of the aerial robots with respect to a non-cooperative dynamic target in GPS-denied environments. The general idea is to use a priori unknown static natural landmarks, randomly distributed in the environment, as reference points for locating UAVs with respect to a dynamic target moving freely in 3D space. The above objective is achieved using only monocular measurements of the target and landmarks, as well as measurements of altitude differential among UAVs. The proposed scheme by authors does not need any geometric knowledge of the moving target nor the landmarks. In addition, there are no assumptions about some cooperative behavior of the target.

The configuration of the proposed cooperative visual-based SLAM system is based on the standard EKF-SLAM (Extended Kalman Filter SLAM) methodology. In this context, it is extremely important to provide the proposed system with properties such as observability and robustness to ill-constrained initial conditions. The above properties have a fundamental role in the convergence of the filter, as shown in [47,48]. Therefore, an extensive nonlinear observability test is carried out in order to analyze the system. In this case, novel, contributive and important theoretical results are presented from this analysis.

In this work, two innovations are proposed to improve the observability properties of the system, and thus its performance. Firstly, the use of a target-centric SLAM configuration is proposed instead of

the use of a more common World-centric or Robot-centric SLAM configuration. In the target-centric SLAM configuration, the system state is parameterized with respect to the target position. Secondly, measurements of altitude differential between pairs of UAVs obtained from altimeters are integrated into the system.

Another key difference from most related works is presented; typically, those works focus only on addressing one side of the dual nature of the estimation-control problem. In this work, additionally to the proposed SLAM estimation technique, a flight formation control scheme based on a back-stepping approach [49] is also proposed, to allow carrying out the formation of quadcopters with respect to the target. In this case, the stability of control laws is proved using the Lyapunov theory. In simulations, the state estimated by the SLAM system is used as a feedback to the proposed control scheme to test the closed-loop performance of both estimator and control.

1.3. Paper Outline

The document is organized in the following manner: Section 2 presents the general system specifications and mathematical models. Section 3 presents the nonlinear observability analysis. In Section 4, the proposed SLAM method is described. Section 5 presents the proposed control system. Section 6 shows the numerical simulations results, and finally, in Section 7, the conclusions and final remarks of this work are given.

2. System Specification

In this section, the mathematical models used in this work are introduced. Firstly, the model used for representing the dynamics of the relative position of the *j*-th UAV-camera system with respect to the target is described. Then, the model representation of the relative position of the *i*-th landmark with respect to the target is described. In addition, the measurement models used in this work are presented: (i) the camera projection model and (ii) the relative altitude model. Finally, the dynamic model of a quadcopter is presented as well.

In applications like aerial vehicles, the attitude and heading (roll, pitch, and yaw) estimation is well handled by available systems (e.g., [50,51]). In particular, in this work, it is assumed that the orientation of the camera is always pointing toward the ground. Considering the above assumption, the system state can be simplified by removing the variables related to attitude and heading (which are provided by the AHRS). In addition, therefore, the problem can be focused on relative position estimation. In addition, it is important to note that, with this assumption, the mathematical formulation is kept simple enough to make the observability analysis of Section 3 a tractable problem. In practice, the looking-downward camera assumption can be easily addressed, for instance, with the use of a servo-controlled camera gimbal.

Regarding the visual sensors required for implementing the proposed system, the only assumption is that a set of visual features of the environment, and one visual feature of the target, can be detected and tracked consistently using any available computer vision algorithm.

2.1. Dynamics of the System

Let consider the following continuous-time model describing the dynamics of the proposed system (see Figure 1):

j-th UAV-camera system



Figure 1. Coordinate reference systems.

$$\dot{\mathbf{x}} = \begin{bmatrix} \mathbf{t} \dot{\mathbf{x}}_{\mathbf{c}}^{\ 1} \\ \mathbf{t} \dot{\mathbf{v}}_{\mathbf{c}}^{\ 1} \\ \vdots \\ \mathbf{t} \dot{\mathbf{x}}_{\mathbf{c}}^{\ j} \\ \mathbf{t} \dot{\mathbf{x}}_{\mathbf{c}}^{\ j} \\ \mathbf{t} \dot{\mathbf{x}}_{\mathbf{c}}^{\ j} \\ \mathbf{t} \dot{\mathbf{x}}_{\mathbf{c}}^{\ j} \\ \mathbf{t} \dot{\mathbf{x}}_{\mathbf{c}}^{\ 1} \\ \vdots \\ \mathbf{t} \dot{\mathbf{x}}_{\mathbf{c}}^{\ 1} \\ \mathbf{t} \dot{\mathbf{x}}_{\mathbf{a}}^{\ 1} \\ \vdots \\ \mathbf{t} \dot{\mathbf{x}}_{\mathbf{a}}^{\ 1} \end{bmatrix} = \begin{bmatrix} \mathbf{x}_{\mathbf{c}}^{\ 1} - \mathbf{v}_{\mathbf{t}} \\ \mathbf{v}_{\mathbf{c}}^{\ j} - \dot{\mathbf{v}}_{\mathbf{t}} \\ \mathbf{v}_{\mathbf{c}}^{\ j} - \mathbf{v}_{\mathbf{t}} \\ \mathbf{0}_{3 \times 1} \\ \mathbf{0}_{3 \times 1} \\ \mathbf{0}_{3 \times 1} \\ \mathbf{0}_{3 \times 1} \\ \mathbf{v}_{\mathbf{a}}^{\ 1} - \mathbf{v}_{\mathbf{t}} \\ \vdots \\ \mathbf{v}_{\mathbf{a}}^{\ i} - \mathbf{v}_{\mathbf{t}} \end{bmatrix} = \begin{bmatrix} \mathbf{t} \mathbf{v}_{\mathbf{c}}^{\ 1} \\ \mathbf{0}_{3 \times 1} \\ \vdots \\ \mathbf{v}_{\mathbf{c}}^{\ j} - \mathbf{v}_{\mathbf{t}} \\ \mathbf{0}_{3 \times 1} \\ \mathbf{v}_{\mathbf{a}}^{\ 1} - \mathbf{v}_{\mathbf{t}} \\ \vdots \\ \mathbf{v}_{\mathbf{a}}^{\ i} - \mathbf{v}_{\mathbf{t}} \end{bmatrix} = \begin{bmatrix} \mathbf{t} \mathbf{v}_{\mathbf{c}}^{\ 1} \\ \mathbf{0}_{3 \times 1} \\ \mathbf{0}_{3 \times 1} \\ \mathbf{0}_{3 \times 1} \\ \mathbf{0}_{3 \times 1} \\ - \mathbf{v}_{\mathbf{t}} \\ \vdots \\ - \mathbf{v}_{\mathbf{t}} \end{bmatrix}$$
(1)

where the state vector **x** is defined by:

$$\mathbf{x} = \begin{bmatrix} \mathbf{t}_{\mathbf{x}_{c}}^{1} & \mathbf{t}_{\mathbf{v}_{c}}^{1} & \cdots & \mathbf{t}_{\mathbf{x}_{c}}^{j} & \mathbf{t}_{\mathbf{v}_{c}}^{j} & \mathbf{v}_{t} & \mathbf{t}_{\mathbf{a}_{a}}^{1} & \cdots & \mathbf{t}_{\mathbf{a}_{a}}^{i} \end{bmatrix}^{\mathrm{T}}$$
(2)

with $i = 1, ..., n_1$ and $j = 1, ..., n_2$, where n_1 and n_2 are respectively the number of landmarks included into the map and the number of UAV-camera systems.

Additionally, let $\mathbf{x}_{t} = \begin{bmatrix} x_{t} & y_{t} & z_{t} \end{bmatrix}^{T}$ represent the position (in meters) of the target, with respect to the reference system *W*. Let $\mathbf{x}_{c}^{j} = \begin{bmatrix} x_{c}^{j} & y_{c}^{j} & z_{c}^{j} \end{bmatrix}^{T}$ represent the position (in meters) of the reference system *C* of the *j*-th camera, with respect to the reference system *W*. Let $\mathbf{v}_{t} = \begin{bmatrix} \dot{x}_{t} & \dot{v}_{t} & \dot{z}_{t} \end{bmatrix}^{T}$ represent the linear velocity (in $\frac{m}{s}$) of the target, with respect to the reference system *W*. Let $\mathbf{v}_{c}^{j} = \begin{bmatrix} \dot{x}_{c}^{j} & \dot{y}_{c}^{j} & \dot{z}_{c}^{j} \end{bmatrix}^{T}$ represent the linear velocity (in $\frac{m}{s}$) of the target, with respect to the reference system *W*. Let $\mathbf{v}_{c}^{j} = \begin{bmatrix} \dot{x}_{a}^{j} & \dot{y}_{a}^{j} & z_{a}^{j} \end{bmatrix}^{T}$ be the position (in meters) of the *i*-th landmark with respect to the reference system *W*. Let $\mathbf{v}_{a}^{i} = \begin{bmatrix} x_{a}^{i} & y_{a}^{i} & z_{a}^{i} \end{bmatrix}^{T}$ be the position (in meters) of the *i*-th landmark with respect to the reference system *W*, defined by its Euclidean parameterization. Let $\mathbf{v}_{a}^{i} = \begin{bmatrix} \dot{x}_{a}^{i} & \dot{y}_{a}^{i} & \dot{z}_{a}^{i} \end{bmatrix}^{T}$ represent the linear velocity (in $\frac{m}{s}$) of the *i*-th landmark, with respect to the reference system *W*. Let $\mathbf{t}_{a}^{j} = \begin{bmatrix} t x_{a}^{j} & t y_{a}^{j} & t z_{a}^{j} \end{bmatrix}^{T}$ represent the linear velocity (in $\frac{m}{s}$) of the *i*-th landmark with respect to the reference system *W*. Let $\mathbf{t}_{a}^{j} = \begin{bmatrix} t x_{a}^{j} & t y_{a}^{j} & t z_{a}^{j} \end{bmatrix}^{T}$ represent the linear velocity (in $\frac{m}{s}$) of the *i*-th landmark, with respect to the reference system *W*. Let $\mathbf{t}_{a}^{j} = \begin{bmatrix} t x_{a}^{j} & t y_{a}^{j} & t z_{a}^{j} \end{bmatrix}^{T}$ represent the relative position (in meters) of the *j*-th camera with respect to the reference system *W*. Let

to the reference system *T*. Let ${}^{t}\mathbf{x_{a}}^{i} = \begin{bmatrix} {}^{t}x_{a}^{i} {}^{t}y_{a}^{i} {}^{t}z_{a}^{i} \end{bmatrix}^{T}$ represent the relative position (in meters) of the *i*-th landmark with respect to the reference system *T*. Finally, let ${}^{t}\mathbf{v_{c}}^{j} = \begin{bmatrix} {}^{t}\dot{x_{c}}^{j} {}^{t}\dot{y_{c}}^{j} {}^{t}\dot{z_{c}} \end{bmatrix}^{T}$ represent the relative linear velocity (in $\frac{m}{s}$) of the *j*-th camera with respect to the reference system *T*. Figure 1 illustrates the coordinate reference systems used in this work. In Equation (1), each UAV-camera, as well as the target, is assumed to move freely in the three-dimensional space. Let note that a non-acceleration kinematic model is assumed for the UAV-camera systems and the target. In addition, note that the landmarks are assumed to remain static. Given the above $\mathbf{v_{a}}^{i} = \mathbf{0}$.

2.2. Camera Measurement Model for Landmarks

Let us consider the projection of a single landmark over the image plane of a camera. Using the pinhole model [52], the following expression is defined:

$${}^{i}\mathbf{z_{c}}{}^{j} = {}^{i}\mathbf{h_{c}}{}^{j} = \begin{bmatrix} {}^{i}u_{c}{}^{j}\\ {}^{i}v_{c}{}^{j} \end{bmatrix} = \frac{1}{{}^{i}z_{d}{}^{j}} \begin{bmatrix} \frac{f_{c}{}^{j}}{d_{u}^{j}} & 0\\ 0 & \frac{f_{c}{}^{j}}{d_{v}^{j}} \end{bmatrix} \begin{bmatrix} {}^{i}x_{d}{}^{j}\\ {}^{i}y_{d}^{j} \end{bmatrix} + \begin{bmatrix} {}^{c}u_{u}{}^{j}+d_{ur}{}^{j}+d_{ut}{}^{j}\\ {}^{c}v_{v}{}^{j}+d_{vr}{}^{j}+d_{vt}{}^{j} \end{bmatrix}$$
(3)

Let ${}^{i}u_{c}^{j}$ and ${}^{i}v_{c}^{j}$ define the coordinates (in pixels) of the projection of the *i*-th landmark over the image of the *j*-th camera. Let f_{c}^{j} be the focal length (in meters) of the *j*-th camera. Let d_{u}^{j} and d_{v}^{j} be the conversion parameters (in *m* / *pixel*) for the *j*-th camera. Let c_{u}^{j} and c_{v}^{j} be the coordinates (in pixels) of the image central point of the *j*-th camera. Let d_{ur}^{j} and d_{vr}^{j} be components (in pixels) accounting for the radial distortion of the *j*-th camera. Let d_{ut}^{j} and d_{vt}^{j} be components (in pixels) accounting for the tangential distortion of the *j*-th camera. All the intrinsic parameters of the *j*-th camera are assumed to be known by means of some calibration method. Let ${}^{i}\mathbf{p}_{d}{}^{j} = \begin{bmatrix} {}^{i}x_{d}^{j} & {}^{i}y_{d}^{j} & {}^{i}z_{d}^{j} \end{bmatrix}^{\mathrm{T}}$ represent the position (in meters) of the *i*-th landmark with respect to the coordinate reference system *C* of the *j*-th camera. Additionally

Additionally,

$${}^{i}\mathbf{p}_{\mathbf{d}}{}^{j} = {}^{\mathbf{W}}\mathbf{R}_{\mathbf{c}}{}^{j}\left({}^{\mathbf{t}}\mathbf{x}_{\mathbf{a}}{}^{i} - {}^{\mathbf{t}}\mathbf{x}_{\mathbf{c}}{}^{j}\right)$$
(4)

where ${}^{W}\mathbf{R}_{\mathbf{c}}{}^{j} \in SO3$ is the rotation matrix that transforms from the world coordinate reference system W to the coordinate reference system C of the *j*-th camera. Recall that the rotation matrix ${}^{W}\mathbf{R}_{\mathbf{c}}{}^{j}$ is known and constant assuming the use of the servo-controlled camera gimbal.

2.3. Camera Measurement Model for the Target

Let consider the projection of the target over the image plane of a camera. In this case, it is assumed that some visual feature points can be extracted from the target by means of some available computer vision algorithm like [53–58]. Using the pinhole model, the following expression is defined:

$${}^{\mathbf{t}}\mathbf{z_{c}}^{j} = {}^{\mathbf{t}}\mathbf{h_{c}}^{j} = \begin{bmatrix} {}^{t}u_{c}^{j} \\ {}^{t}v_{c}^{j} \end{bmatrix} = \frac{1}{{}^{t}z_{d}^{j}} \begin{bmatrix} \frac{f_{c}^{j}}{d_{u}^{j}} & 0 \\ 0 & \frac{f_{c}^{j}}{d_{v}^{j}} \end{bmatrix} \begin{bmatrix} {}^{t}x_{d}^{j} \\ {}^{t}y_{d}^{j} \end{bmatrix} + \begin{bmatrix} c_{u}^{j} + d_{ur}^{j} + d_{ut}^{j} \\ c_{v}^{j} + d_{vr}^{j} + d_{vt}^{j} \end{bmatrix}$$
(5)

Let ${}^{t}\mathbf{p}_{d}{}^{j} = \begin{bmatrix} {}^{t}x_{d}^{j} & {}^{t}y_{d}^{j} & {}^{t}z_{d}^{j} \end{bmatrix}^{T}$ represent the position (in meters) of the target with respect to the coordinate reference system *C* of the *j*-th camera. Additionally,

$${}^{t}\mathbf{p}_{d}{}^{j} = {}^{W}\mathbf{R}_{c}{}^{j}\left(-{}^{t}\mathbf{x}_{c}{}^{j}\right)$$
(6)

2.4. Altitude Differential Measurement Model

In this work, the altitude differential between a pair of UAVs will be used as a filter update measurement. In this case, the altitude differential between the *j*-th UAV and the *n*-th UAV, given by a pair of on-board altimeters in each vehicle, is defined by:

$${}^{n}z_{a}^{j} = {}^{n}h_{a}^{j} = {}^{t}z_{c}^{n} - {}^{t}z_{c}^{j}$$
⁽⁷⁾

2.5. Dynamic Model of the Quadcopter

This section presents the dynamic model of a quadcopter which is composed of a rigid structure and four rotors. The plant has six degrees of freedom: three for the translation and three for the rotation (see Figure 1). Using the Euler–Lagrange formalism and the parametrization with respect to the Tait–Bryan angles, the model can be defined similarly as [59]:

$$\ddot{\mathbf{x}}_{\mathbf{q}}^{j} = \begin{bmatrix} \ddot{x}_{q}^{j} \\ \ddot{y}_{q}^{j} \\ \ddot{z}_{q}^{j} \end{bmatrix} = \frac{\mu^{j}}{m^{j}} \begin{bmatrix} \sin(\psi^{j})\sin(\phi^{j}) + \cos(\psi^{j})\sin(\theta^{j})\cos(\phi^{j}) \\ \sin(\psi^{j})\sin(\theta^{j})\cos(\phi^{j}) - \cos(\psi^{j})\sin(\phi^{j}) \\ \cos(\theta^{j})\cos(\phi^{j}) - \frac{g\cdot m^{j}}{u_{q}^{j}\cdot\mu^{j}} \end{bmatrix} u_{q}^{j}$$
(8)

$$\ddot{\boldsymbol{\sigma}}^{j} = \begin{bmatrix} \ddot{\boldsymbol{\phi}}^{j} \\ \ddot{\boldsymbol{\theta}}^{j} \\ \ddot{\boldsymbol{\psi}}^{j} \end{bmatrix} = \mathbf{g}^{j} + \mathbf{B}^{j} \boldsymbol{\tau}^{j} = \begin{bmatrix} \left(\frac{I_{y}^{j} - I_{z}^{j}}{I_{x}^{j}} \right) \dot{\boldsymbol{\theta}}^{j} \dot{\boldsymbol{\psi}}^{j} + \left(\frac{I_{p}^{j}}{I_{x}^{j}} \right) \dot{\boldsymbol{\theta}}^{j} \Omega^{j} \\ \left(\frac{I_{x}^{j} - I_{y}^{j}}{I_{y}^{j}} \right) \dot{\boldsymbol{\phi}}^{j} \dot{\boldsymbol{\psi}}^{j} + \left(\frac{I_{p}^{j}}{I_{y}^{j}} \right) \dot{\boldsymbol{\phi}}^{j} \Omega^{j} \\ \left(\frac{I_{x}^{j} - I_{y}^{j}}{I_{z}^{j}} \right) \dot{\boldsymbol{\phi}}^{j} \dot{\boldsymbol{\theta}}^{j} d^{j} \end{bmatrix} + \begin{bmatrix} \frac{\mu^{j}}{I_{x}^{j}} & 0 & 0 \\ 0 & \frac{\mu^{j}}{I_{y}^{j}} & 0 \\ 0 & 0 & \frac{d^{j}}{I_{z}^{j}} \end{bmatrix} \begin{bmatrix} \tau_{\theta}^{j} \\ \tau_{\theta}^{j} \\ \tau_{\psi}^{j} \end{bmatrix}$$
(9)

Let $\mathbf{x}_{\mathbf{q}}^{j} = [x_{q}^{j}, y_{q}^{j}, z_{q}^{j}]^{\mathrm{T}}$ be the position (in meters) of the *j*-th quadcopter, with respect to the reference system *W*. Let $\sigma^{j} = [\phi^{j}, \theta^{j}, \psi^{j}]^{\mathrm{T}}$ be the Tait–Bryan angles (in radians) of the *j*-th quadcopter, with respect to the reference system *W*. Let m^{j} be the mass (in kg) of the *j*-th quadcopter. Let μ^{j} be the thrust factor (in N · s²) of the *j*-th quadcopter. Let *g* be the constant of gravity (in m/s²). Let u_{q}^{j} be the total force (in *N*) of the *j*-th quadcopter, supplied by the four rotors over the axis *z* with respect to the coordinate reference system *Q*. Let J_{p}^{j} be total inertia moment (in N · m · s²) of the *j*-th quadcopter, due to the rotors. Let d^{j} be the drag factor (in N · m · s²) of the *j*-th quadcopter. Let I_{x}^{j} , I_{y}^{j} , and I_{z}^{j} be the inertia moment over each axis (in N · m · s²) of the *j*-th quadcopter. Let τ_{ϕ}^{j} , τ_{θ}^{j} , and τ_{ψ}^{j} be the torque (in $N \cdot m$) of the *j*-th quadcopter, supplied by the rotors over each axis. Finally, let Ω^{j} be the overall propeller rotational speed (in rad/s) of the *j*-th quadcopter.

The inputs are defined by:

$$\begin{bmatrix} u_{q}^{j} \\ \tau_{\phi}^{j} \\ \tau_{\theta}^{j} \\ \tau_{\psi}^{j} \end{bmatrix} = \begin{bmatrix} 1 & 1 & 1 & 1 \\ -l^{j} & 0 & l^{j} & 0 \\ 0 & l^{j} & 0 & -l^{j} \\ 1 & -1 & 1 & -1 \end{bmatrix} \begin{bmatrix} f_{1}^{j} \\ f_{2}^{j} \\ f_{3}^{j} \\ f_{4}^{j} \end{bmatrix}$$
(10)

where

$$f_m^j = \left(w_m^j\right)^2 \tag{11}$$

$$\Omega^{j} = w_{1}^{j} - w_{2}^{j} + w_{3}^{j} - w_{4}^{j}$$
(12)

In addition, $m = \{1, ..., 4\}$. Let l^j be the distance (in *m*) from the center of mass of the *j*-th quadcopter to the center of one rotor. Let f_m^j be the thrust force (in *N*), supplied for the *m* rotor of the *j*-th quadcopter. In addition, let w_m^j be the angular velocity (in rad/s) of the *m* rotor of the *j*-th quadcopter.

3. Observability Analysis

In this section, the nonlinear observability properties of the proposed system are studied. Observability is an inherent property of a dynamic system and has an important role in the accuracy and stability of its estimation process; moreover, this fact has important consequences in the context of SLAM and the convergence of the EKF.

A system is defined as observable if the initial state \mathbf{x}_0 , at any initial time t_0 , can be determined given the state transition $\dot{\mathbf{x}} = \mathbf{f}(\mathbf{x})$, the observation model $\mathbf{y} = \mathbf{h}(\mathbf{x})$ of the system, and observations $\mathbf{z}[t_0, t]$ from time t_0 to a finite time t. In Hermann and Krener [60], it is demonstrated that a nonlinear system is *locally weakly observable* if the observability rank condition $rank(\mathcal{O}) = dim(\mathbf{x})$ is verified, where \mathcal{O} is the observability matrix.

3.1. Observability Matrix

The observability matrix \mathcal{O} can be computed from:

$$\boldsymbol{\mathcal{O}} = \begin{bmatrix} \frac{\partial \left(\mathcal{L}_{\mathbf{f}}^{0}({}^{i}\mathbf{h}_{\mathbf{c}}^{j})\right)}{\partial \mathbf{x}} & \frac{\partial \left(\mathcal{L}_{\mathbf{f}}^{1}({}^{i}\mathbf{h}_{\mathbf{c}}^{j})\right)}{\partial \mathbf{x}} & \dots & \frac{\partial \left(\mathcal{L}_{\mathbf{f}}^{0}\left({}^{t}\mathbf{h}_{\mathbf{c}}^{j}\right)\right)}{\partial \mathbf{x}} & \frac{\partial \left(\mathcal{L}_{\mathbf{f}}^{1}\left({}^{t}\mathbf{h}_{\mathbf{c}}^{j}\right)\right)}{\partial \mathbf{x}} & \dots & \frac{\partial \left(\mathcal{L}_{\mathbf{f}}^{1}\left({}^{n}\mathbf{h}_{a}^{j}\right)\right)}{\partial \mathbf{x}} & \dots \end{bmatrix}^{\mathrm{T}}$$
(13)

where $\mathcal{L}_{f}^{s}\mathbf{h}$ is the *s*-th-order Lie Derivative [61], of the scalar field \mathbf{h} with respect to the vector field \mathbf{f} . In this work, for computing Equation (13), zero-order and first-order Lie derivatives will be used for each kind of measurement.

In case of the measurements given by a monocular camera, according to Equations (3) and (1), the following zero-order Lie derivative is defined for the projections of landmarks:

$$\frac{\partial \left(\mathcal{L}_{\mathbf{f}}^{0}\left({}^{i}\mathbf{h}_{\mathbf{c}}^{j}\right)\right)}{\partial \mathbf{x}} = \begin{bmatrix} \mathbf{0}_{2\times 6(j-1)} & -{}^{i}\mathbf{H}_{\mathbf{c}}^{j} \cdot {}^{\mathbf{W}}\mathbf{R}_{\mathbf{c}}^{j} & \mathbf{0}_{2\times 3} & \mathbf{0}_{2\times 6(n_{2}-j)} & | & \mathbf{0}_{2\times 3} & | & \mathbf{0}_{2\times 3(i-1)} & {}^{i}\mathbf{H}_{\mathbf{c}}^{j} \cdot {}^{\mathbf{W}}\mathbf{R}_{\mathbf{c}}^{j} & \mathbf{0}_{2\times 3(n_{1}-i)} \end{bmatrix}$$
(14)

where

$${}^{i}\mathbf{H_{c}}^{j} = \frac{f_{c}^{j}}{\left({}^{i}z_{d}^{j}\right)^{2}} \begin{bmatrix} \frac{{}^{i}z_{d}^{j}}{d_{u}^{j}} & 0 & -\frac{{}^{i}x_{d}^{j}}{d_{u}^{j}} \\ 0 & \frac{{}^{i}z_{d}^{j}}{d_{v}^{j}} & -\frac{{}^{i}y_{d}^{j}}{d_{v}^{j}} \end{bmatrix}$$
(15)

For the same kind of measurement, the following first-order Lie derivative can be defined:

$$\frac{\partial \left(\mathcal{L}_{\mathbf{f}}^{1}\left(^{i}\mathbf{h}_{\mathbf{c}}^{j}\right)\right)}{\partial \mathbf{x}} = \begin{bmatrix} \mathbf{0}_{2\times 6(j-1)} & ^{i}\mathbf{H}_{\mathbf{d}\mathbf{c}}^{j} & -^{i}\mathbf{H}_{\mathbf{c}}^{j} \cdot ^{\mathbf{W}}\mathbf{R}_{\mathbf{c}}^{j} & \mathbf{0}_{2\times 6(n_{2}-j)} & | & ^{i}\mathbf{H}_{\mathbf{c}}^{j} \cdot ^{\mathbf{W}}\mathbf{R}_{\mathbf{c}}^{j} & | & \mathbf{0}_{2\times 3(i-1)} & -^{i}\mathbf{H}_{\mathbf{d}\mathbf{c}}^{j} & \mathbf{0}_{2\times 3(n_{1}-i)} \end{bmatrix}$$
(16)

where

$${}^{i}\mathbf{H}_{dc}{}^{j} = \begin{bmatrix} {}^{i}\mathbf{H}_{1}{}^{j} {}^{i}\mathbf{H}_{2}{}^{j} {}^{i}\mathbf{H}_{3}{}^{j} \end{bmatrix} \begin{pmatrix} {}^{\mathbf{W}}\mathbf{R}_{c}{}^{j} \end{pmatrix}^{2} \begin{pmatrix} {}^{t}\mathbf{v}_{c}{}^{j} + \mathbf{v}_{t} \end{pmatrix}$$
(17)

and

$${}^{i}\mathbf{H}_{1}{}^{j} = \frac{f_{c}^{j}}{d_{u}^{j} \left(iz_{d}^{j}\right)^{2}} \begin{bmatrix} 0 & 0 & -1 \\ 0 & 0 & 0 \end{bmatrix} \qquad {}^{i}\mathbf{H}_{2}{}^{j} = \frac{f_{c}^{j}}{d_{v}^{j} \left(iz_{d}^{j}\right)^{2}} \begin{bmatrix} 0 & 0 & 0 \\ 0 & 0 & -1 \end{bmatrix} \qquad {}^{i}\mathbf{H}_{3}{}^{j} = \frac{f_{c}^{j}}{\left(iz_{d}^{j}\right)^{3}} \begin{bmatrix} -\frac{iz_{d}^{j}}{d_{u}^{j}} & 0 & \frac{2\left(ix_{d}^{j}\right)}{d_{u}^{j}} \\ 0 & -\frac{iz_{d}^{j}}{d_{v}^{j}} & \frac{2\left(iy_{d}^{j}\right)}{d_{v}^{j}} \end{bmatrix}$$
(18)

In case of the measurement given by a monocular camera, according to Equatons (5) and (1), the following zero-order Lie derivative is defined for the projections of the target:

$$\frac{\partial \left(\mathcal{L}_{\mathbf{f}}^{0} \left({}^{\mathbf{t}} \mathbf{h}_{\mathbf{c}}^{j} \right) \right)}{\partial \mathbf{x}} = \begin{bmatrix} \mathbf{0}_{2 \times 6(j-1)} & -{}^{\mathbf{t}} \mathbf{H}_{\mathbf{c}}^{j} \cdot {}^{\mathbf{W}} \mathbf{R}_{\mathbf{c}}^{j} & \mathbf{0}_{2 \times 3} & \mathbf{0}_{2 \times 6(n_{2}-j)} & | & \mathbf{0}_{2 \times 3} & | & \mathbf{0}_{2 \times 3n_{1}} \end{bmatrix}$$
(19)

where

$${}^{\mathbf{t}}\mathbf{H}_{\mathbf{c}}^{\ j} = \frac{f_{c}^{j}}{\left({}^{t}z_{d}^{j}\right)^{2}} \begin{bmatrix} \frac{{}^{t}z_{d}^{j}}{d_{u}^{j}} & 0 & -\frac{{}^{t}x_{d}^{j}}{d_{u}^{j}} \\ 0 & \frac{{}^{t}z_{d}^{j}}{d_{v}^{j}} & -\frac{{}^{t}y_{d}^{j}}{d_{v}^{j}} \end{bmatrix}$$
(20)

For the same kind of measurement, the following first-order Lie derivative can be defined:

$$\frac{\partial \left(\mathcal{L}_{\mathbf{f}}^{1} \left({}^{\mathbf{t}} \mathbf{h}_{\mathbf{c}}^{j} \right) \right)}{\partial \mathbf{x}} = \begin{bmatrix} \mathbf{0}_{2 \times 6(j-1)} & {}^{\mathbf{t}} \mathbf{H}_{\mathbf{dc}}^{j} & -{}^{\mathbf{t}} \mathbf{H}_{\mathbf{c}}^{j} \cdot {}^{\mathbf{W}} \mathbf{R}_{\mathbf{c}}^{j} & \mathbf{0}_{2 \times 6(n_{2}-j)} & | & \mathbf{0}_{2 \times 3} & | & \mathbf{0}_{2 \times 3n_{1}} \end{bmatrix}$$
(21)

where

$${}^{\mathbf{t}}\mathbf{H}_{\mathbf{dc}}{}^{j} = \begin{bmatrix} {}^{\mathbf{t}}\mathbf{H}_{\mathbf{1}}{}^{j} {}^{\mathbf{t}}\mathbf{H}_{\mathbf{2}}{}^{j} {}^{\mathbf{t}}\mathbf{H}_{\mathbf{3}}{}^{j} \end{bmatrix} \begin{pmatrix} {}^{\mathbf{W}}\mathbf{R}_{\mathbf{c}}{}^{j} \end{pmatrix}^{2} \begin{pmatrix} {}^{\mathbf{t}}\mathbf{v}_{\mathbf{c}}{}^{j} \end{pmatrix}$$
(22)

and

$${}^{\mathbf{t}}\mathbf{H}_{\mathbf{1}}{}^{j} = \frac{f_{c}^{j}}{d_{u}^{j} \left({}^{t}z_{d}^{j}\right)^{2}} \begin{bmatrix} 0 & 0 & -1 \\ 0 & 0 & 0 \end{bmatrix} \qquad {}^{\mathbf{t}}\mathbf{H}_{\mathbf{2}}{}^{j} = \frac{f_{c}^{j}}{d_{v}^{j} \left({}^{t}z_{d}^{j}\right)^{2}} \begin{bmatrix} 0 & 0 & 0 \\ 0 & 0 & -1 \end{bmatrix} \qquad {}^{\mathbf{t}}\mathbf{H}_{\mathbf{3}}{}^{j} = \frac{f_{c}^{j}}{\left({}^{t}z_{d}^{j}\right)^{3}} \begin{bmatrix} -\frac{{}^{t}z_{d}^{j}}{d_{u}^{l}} & 0 & \frac{2\left({}^{t}x_{d}^{j}\right)}{d_{u}^{l}} \\ 0 & -\frac{{}^{t}z_{d}^{j}}{d_{v}^{j}} & \frac{2\left({}^{t}y_{d}^{j}\right)}{d_{v}^{l}} \end{bmatrix}$$
(23)

In case of the measurement of the altitude differential, according to Equations (7) and (1), for the zero-order Lie derivative, if j < n (the index of the *j*-th camera is lesser than the index of the *n*-th camera)

$$\frac{\partial \left(\mathcal{L}_{\mathbf{f}}^{0} \left({}^{n} h_{a}^{j} \right) \right)}{\partial \mathbf{x}} = \begin{bmatrix} \mathbf{0}_{1 \times 6(j-1)} & -\mathbf{M}_{\mathbf{x}} & \mathbf{0}_{1 \times 6(n-j-1)} & \mathbf{M}_{\mathbf{x}} & \mathbf{0}_{1 \times 6(n_{2}-n)} & | & \mathbf{0}_{1 \times 3} & | & \mathbf{0}_{1 \times 3n_{1}} \end{bmatrix}$$
(24)

On the other hand, if j > n (the index of the *j*-th camera is higher than the index of the *n*-th camera), then

$$\frac{\partial \left(\mathcal{L}_{\mathbf{f}}^{0} \left({}^{n} h_{a}^{j} \right) \right)}{\partial \mathbf{x}} = \begin{bmatrix} \mathbf{0}_{1 \times 6(n-1)} & \mathbf{M}_{\mathbf{x}} & \mathbf{0}_{1 \times 6(j-n-1)} & -\mathbf{M}_{\mathbf{x}} & \mathbf{0}_{1 \times 6(n_{2}-j)} & | & \mathbf{0}_{1 \times 3} & | & \mathbf{0}_{1 \times 3n_{1}} \end{bmatrix}$$
(25)

and for both cases

$$\mathbf{M}_{\mathbf{x}} = \begin{bmatrix} \mathbf{0}_{1 \times 2} & 1 & \mathbf{0}_{1 \times 3} \end{bmatrix}$$
(26)

For the first-order Lie derivative, if j < n:

$$\frac{\partial \left(\mathcal{L}_{\mathbf{f}}^{1} \left({}^{n} h_{a}^{j} \right) \right)}{\partial \mathbf{x}} = \begin{bmatrix} \mathbf{0}_{1 \times 6(j-1)} & -\mathbf{M}_{\mathbf{dx}} & \mathbf{0}_{1 \times 6(n-j-1)} & \mathbf{M}_{\mathbf{dx}} & \mathbf{0}_{1 \times 6(n_{2}-n)} & | & \mathbf{0}_{1 \times 3} & | & \mathbf{0}_{1 \times 3n_{1}} \end{bmatrix}$$
(27)

In addition, if j > n:

$$\frac{\partial \left(\mathcal{L}_{\mathbf{f}}^{1} \left({}^{n} h_{a}^{j} \right) \right)}{\partial \mathbf{x}} = \begin{bmatrix} \mathbf{0}_{1 \times 6(n-1)} & \mathbf{M}_{\mathbf{dx}} & \mathbf{0}_{1 \times 6(j-n-1)} & -\mathbf{M}_{\mathbf{dx}} & \mathbf{0}_{1 \times 6(n_{2}-j)} & | & \mathbf{0}_{1 \times 3} & | & \mathbf{0}_{1 \times 3n_{1}} \end{bmatrix}$$
(28)

with

$$\mathbf{M}_{\mathbf{dx}} = \begin{bmatrix} \mathbf{0}_{1 \times 5} & 1 \end{bmatrix}$$
(29)

	$0_{2\times 6(j-1)} \qquad -^{i}\mathbf{H}_{\mathbf{c}}^{j} \cdot \mathbf{W}\mathbf{R}_{\mathbf{c}}^{j} 0_{2\times 3}$ $0_{2\times 6(j-1)} \qquad ^{i}\mathbf{H}_{\mathbf{d}\mathbf{c}}^{j} -^{i}\mathbf{H}_{\mathbf{c}}^{j} \cdot \mathbf{W}\mathbf{R}_{\mathbf{c}}^{j}$	$0_{2 \times 6(n_2 - j)}$ $0_{2 \times 6(n_2 - j)}$	$0_{2\times 3} \mid \mathbf{H}_{\mathbf{c}}^{j} \cdot \mathbf{W} \mathbf{R}_{\mathbf{c}}^{j} \mid$	$\begin{aligned} 0_{2\times 3(i-1)} & {}^{i}\mathbf{H_{c}}^{j}\cdot\mathbf{^{W}R_{c}}^{j} & 0_{2\times 3(n_{1}-i)} \\ 0_{2\times 3(i-1)} & -{}^{i}\mathbf{H_{dc}}^{j} & 0_{2\times 3(n_{1}-i)} \\ & & \cdot \end{aligned}$	
	: $0_{2\times 6(j-1)}$ $-\mathbf{t}\mathbf{H}_{\mathbf{c}}^{\ j}\cdot\mathbf{W}\mathbf{R}_{\mathbf{c}}^{\ j}$ $0_{2\times 3}$	$0_{2 \times 6(n_2-j)}$: 0 _{2×3}	: 0 _{2×3n1}	
	$0_{2\times 6(j-1)} \qquad {}^{\mathbf{t}}\mathbf{H}_{\mathbf{dc}}{}^{j} - {}^{\mathbf{t}}\mathbf{H}_{\mathbf{c}}{}^{j} \cdot {}^{\mathbf{W}}\mathbf{R}_{\mathbf{c}}{}^{j}$	$0_{2\times 6(n_2-j)} \mid$	0 _{2×3}	$0_{2 \times 3n_1}$	
O =	÷		÷	÷	(30)
	$0_{1 \times 6(j-1)}$ $-\mathbf{M}_{\mathbf{x}}$ $0_{1 \times 6(n-j-1)}$ $\mathbf{M}_{\mathbf{x}}$	$0_{1 \times 6(n_2-n)}$	0 _{1×3}	$0_{1 \times 3n_1}$	
	$0_{1 \times 6(j-1)}$ $-\mathbf{M}_{\mathbf{dx}}$ $0_{1 \times 6(n-j-1)}$ $\mathbf{M}_{\mathbf{dx}}$	$0_{1\times 6(n_2-n)} \mid$	0 _{1×3}	$0_{1 \times 3n_1}$	
	÷		÷	÷	
	$0_{1 \times 6(n-1)}$ $\mathbf{M}_{\mathbf{x}}$ $0_{1 \times 6(j-n-1)}$ $-\mathbf{M}_{\mathbf{x}}$	$0_{1\times 6(n_2-j)} $	0 1×3	$0_{1 \times 3n_1}$	
	$0_{1 \times 6(n-1)} \mathbf{M}_{\mathbf{dx}} 0_{1 \times 6(j-n-1)} -\mathbf{M}_{\mathbf{dx}}$	$0_{1\times 6(n_2-j)} $	0 _{1×3}	$0_{1 \times 3n_1}$	
				÷	

Using the Lie derivatives described above, the observability matrix for the proposed system Equation (1) can be defined as follows:

3.2. Theoretical Results

Three different cases of system configurations were analyzed. The idea is to study how the observability of the system is affected due to the availability (or unavailability) of the different types of measurements. The cases to be considered are:

- Case 1: The following measurements are available: (i) monocular measurements of the projection of landmarks over each UAV-camera system, (ii) multiple (two or more) monocular measurements of the projection of the target over each UAV-camera system.
- Case 2: The following measurements are available: (i) monocular measurements of projection of landmarks over each UAV-camera system, (ii) a single monocular measurement of the projection of the target over a UAV-camera system, and (iii) the altitude differential measurements between UAV-camera systems.
- Case 3: The following measurements are available: (i) monocular measurements of the projection of landmarks over each UAV-camera system, (ii) multiple (two or more) monocular measurements of the projection of the target over each UAV-camera system, and (iii) the altitude differential measurements between UAV-camera systems.

3.2.1. Case 1

For the first case, considering only the respective derivatives on the observability matrix, Equation (30), the maximum rank of the observability matrix \mathcal{O} is $rank(\mathcal{O}) = (3n_1 + 6n_2 + 3) - 1$, where n_1 is the number of landmarks being measured, n_2 is the number of robots, and 3 is the number of states of the linear velocity of the target. In this case, n_1 is multiplied by 3, since this is the number of states per landmark, and n_2 is multiplied by 6, since this is the number of states per robot. Therefore,

O will be rank deficient ($rank(O) < dim(\mathbf{x})$). The unobservable modes are spanned by the right nullspace basis of the observability matrix O:

$$\mathbf{N}_{1} = null(\mathcal{O}) = \begin{bmatrix} \mathbf{t}_{\mathbf{x}_{c}}^{1} & \mathbf{t}_{\mathbf{v}_{c}}^{1} & \cdots & \mathbf{t}_{\mathbf{x}_{c}}^{j} & \mathbf{t}_{\mathbf{v}_{c}}^{j} & | \mathbf{v}_{t} & | & \mathbf{t}_{\mathbf{x}_{a}}^{1} & \cdots & \mathbf{t}_{\mathbf{x}_{a}}^{i} \end{bmatrix}^{\mathrm{T}} = \mathbf{x}$$
(31)

It is straightforward to verify that the right nullspace basis of \mathcal{O} spans for N₁, (i.e., \mathcal{O} N₁ = 0). From Equation (31), it can be seen that the unobservable modes cross through all states; therefore, all states are unobservable. It should be noted that, adding Lie derivatives of higher-order to the observability matrix, the previous result does not improve.

3.2.2. Case 2

For the second case, the maximum rank of the observability matrix \mathcal{O} is $rank(\mathcal{O}) = (3n_1 + 6n_2 + 3) - 2$. Therefore, \mathcal{O} will be rank deficient ($rank(\mathcal{O}) < dim(\mathbf{x})$). In this case, the unobservable modes are spanned by the following right nullspace basis of the observability matrix \mathcal{O} :

$$\mathbf{N}_{2} = null(\boldsymbol{\mathcal{O}}) = \begin{bmatrix} \mathbf{0}_{3\times1} & \mathbf{t}_{\mathbf{x}_{c}^{j}} & \cdots & \mathbf{0}_{3\times1} & \mathbf{t}_{\mathbf{x}_{c}^{j}} & | & \mathbf{t}_{\mathbf{x}_{c}^{j}} & | & \mathbf{0}_{3\times1} & \cdots & \mathbf{0}_{3\times1} \\ \mathbf{t}_{\mathbf{x}_{c}^{j}} & \begin{bmatrix} \mathbf{t}_{\mathbf{x}_{c}^{j}} \times \mathbf{t}_{\mathbf{v}_{c}^{j}} \end{bmatrix}_{\times} \mathbf{z} & \cdots & \mathbf{t}_{\mathbf{x}_{c}^{j}} & \begin{bmatrix} \mathbf{t}_{\mathbf{x}_{c}^{j}} \times \mathbf{t}_{\mathbf{v}_{c}^{j}} \end{bmatrix}_{\times} \mathbf{z} & | & \begin{bmatrix} \mathbf{t}_{\mathbf{x}_{c}^{j}} \times \mathbf{t}_{\mathbf{v}_{c}^{j}} \end{bmatrix}_{\times} \mathbf{z} & | & \mathbf{t}_{\mathbf{x}_{c}^{j}} & \cdots & \mathbf{t}_{\mathbf{x}_{c}^{j}} \end{bmatrix}^{\mathrm{T}}$$
(32)

where $\mathbf{z} = \begin{bmatrix} 0 & 0 & \frac{1}{tz_c^j} \end{bmatrix}^T$. The states ${}^t\mathbf{x_c}{}^j$, ${}^t\mathbf{v_c}{}^j$ involved in the right nullspace basis ($\mathbf{N_2}$) belong to the *j*-th robot that observes the target. It is straightforward to verify that the right nullspace basis of \mathcal{O} spans for $\mathbf{N_2}$, (i.e., $\mathcal{O}\mathbf{N_2} = \mathbf{0}$). From Equation (32) it can be seen that the unobservable modes cross through all states, therefore, all states are unobservable. It should be noted that adding Lie derivatives of higher-order to the observability matrix the previous result does not improve.

3.2.3. Case 3

For the third case, the maximum rank of the observability matrix \mathcal{O} is $rank(\mathcal{O}) = (3n_1 + 6n_2 + 3)$. Given that $dim(\mathbf{x}) = 3n_1 + 6n_2 + 3$, then, the system under the third case is *locally weakly observable* because $rank(\mathcal{O}) = dim(\mathbf{x})$.

3.2.4. Remarks to the Theoretical Results

The results of the observability analysis are summarized in Table 1.

	Monocular Measurements of Landmarks	Monocular Measurements of the Target (Single)	Monocular Measurements of the Target (Multiple)	Altitude Differential Measurements	Unobservable Modes	Unobservable States	Observable States
Case 1	Yes	No	Yes	No	1	x	-
Case 2	Yes	Yes	No	Yes	2	x	-
Case 3	Yes	No	Yes	Yes	0	-	x

Table 1. Results of the nonlinear observability analysis of the proposed system.

- Having altitude differential measurements between two UAV-camera systems is a necessary condition to obtain the previous results (see Figure 2). In this case, results do not improve when adding more altitude differentials.
- It is necessary to have at least two monocular measurements of the target in order to obtain the
 previous results (see Figure 2).
- To obtain the previous results, it is necessary to link the members of the multi-UAV system through at least two measurements in common (see Figure 2). That means that: (i) a robot needs to share the observation of at least two landmarks with any other robot; or (ii) a robot needs to share the observation of at least one landmark and the target with any other robot; or (iii) a robot needs to share the observation of at least one landmark with any other robot having the

measurement of its altitude differential with that same robot; or (iv) a robot needs to share the observation of the target with any other robot having the measurement of its altitude differential with that same robot.



Figure 2. Minimum requirements to obtain the results of the observability analysis for the proposed system in Case 3.

4. EKF-Based SLAM

In this work, the system state in Equation (2) is estimated using the EKF-based SLAM methodology [62,63]. Figure 3 shows the architecture of the proposed system. In this case, it is assumed a ground-based centralized architecture, where all the most computing-demanding processes, like the visual processing and the main estimation process, are carried out on a ground-based computer. Algorithm 1 outlines the proposed EKF-SLAM based method.



Figure 3. Block diagram showing the EKF-SLAM architecture of the proposed system.

Algorithm 1: EKF-SLAM target-Centric Multi-UAV

Data: $\hat{\mathbf{x}}_0$, \mathbf{P}_0 , \mathbf{Q} , \mathbf{R} , ${}^i\mathbf{R}^j$, th_1 , th_2 (where th_1 is an arbitrary threshold for the least number of features allowed within the map. In addition, th_2 is an arbitrary threshold for the greatest number of sample steps allowed without measurements for a map feature) Input: $\hat{\mathbf{x}}_{k-1}$, \mathbf{P}_{k-1} **Output:** $\hat{\mathbf{x}}_k$, \mathbf{P}_k 1 while State estimation is carried out do A. System Prediction $[\hat{\mathbf{x}}_{k}^{-}, \mathbf{P}_{k}^{-}] = EKF_Prediction(\hat{\mathbf{x}}_{k-1}, \mathbf{P}_{k-1})$ 2 B. Measurements Updates B.1. Visual Updates (Landmarks) if Visual measurements of landmarks are available then 3 for 1 to i do 4 if i-th landmark was measured then 5 for 1 to j do 6 if i-th landmark was measured by the j-th UAV then 7 $[\mathbf{\hat{x}}_k, \mathbf{P}_k] = EKF_Update_Visual_Landmarks(\mathbf{\hat{x}}_k^-, \mathbf{P}_k^-, {}^i\mathbf{h_c}^j, {}^i\mathbf{z_c}^j)$ 8 9 end end 10 11 end end 12 13 end B.2. Visual Updates (target) 14 if Visual measurements of the target are available then for 1 to j do 15 if Target was measured by the *j*-th UAV then 16 $\left| [\hat{\mathbf{x}}_k, \mathbf{P}_k] = EKF_Update_Visual_Target(\hat{\mathbf{x}}_k^-, \mathbf{P}_k^-, {}^{\mathbf{t}}\mathbf{h}_c^{j}, {}^{\mathbf{t}}\mathbf{z}_c^{j}) \right|$ 17 end 18 19 end end 20 B.3. Altitude Differential Updates if Altitude Differential measurements between pairs of UAVs are available then 21 22 **for** 1 *to n*₃ **do** (where n_3 is the number of Altitude Differential measurements) $[\mathbf{\hat{x}}_k, \mathbf{P}_k] = EKF_Update_Altitude_Differential(\mathbf{\hat{x}}_k^-, \mathbf{P}_k^-, nh_a^j, nz_a^j)$ 23 end 24 end 25 C. Map features initialization 26 if $n_1 < th_1$ then 27 **for** 1 to n₄ **do** (where n_4 is the number of new map features to initialize) $[\mathbf{x}, \mathbf{P}_{new}] = Initialization_New_Map_Features(\hat{\mathbf{x}}_k, \mathbf{P}_k, {^iu_c^j}, {^iu_c^j}, {^i\mathbf{R}^j})$ 28 end 29 end 30 D. Map management for 1 to i do 31 if ${}^{t}k_{a}^{i} > th_{2}$ then 32 (where ${}^tk^i_a$ is the number of sample steps without measurements for the *i*-th landmark. ${}^{t}x_{a}{}^{i} = [$ 33 $\mathbf{P}(^{\mathbf{t}}\mathbf{x_{a}}^{i}) = []$ 34 35 end end 36 37 end

4.1. System Prediction

At every step k, the estimated system state $\hat{\mathbf{x}}$ takes a step forward by the following discrete model:

$$\mathbf{x}_{k} = \mathbf{f}(\mathbf{x}_{k-1}, \mathbf{n}_{k-1}) = \begin{bmatrix} \mathbf{t}_{\mathbf{x}_{c_{k}}}^{1} \\ \mathbf{t}_{\mathbf{v}_{c_{k}}}^{1} \\ \mathbf{t}_{\mathbf{v}_{c_{k}}}^{1} \\ \mathbf{t}_{\mathbf{v}_{c_{k}}}^{j} \\ \mathbf{t}_{\mathbf{v}_{c_{k}}}^{j} \\ \mathbf{t}_{\mathbf{v}_{c_{k}}}^{1} \\ \mathbf{t}_{\mathbf{v}_{c_{k}}}^{1} + (\mathbf{t}_{\mathbf{v}_{c_{k-1}}}^{1}) \Delta t \\ \mathbf{t}_{\mathbf{v}_{c_{k-1}}}^{j} + (\mathbf{t}_{\mathbf{v}_{c_{k-1}}}^{j}) \Delta t \\ \mathbf{t}_{\mathbf{v}_{c_{k-1}}}^{j} + (\mathbf{t}_{\mathbf{v}_{c_{k-1}}}^{j}) \Delta t \\ \mathbf{t}_{\mathbf{v}_{c_{k-1}}}^{j} + \mathbf{t}_{\mathbf{v}_{c_{k-1}}}^{j} \\ \mathbf{t}_{\mathbf{v}_{c_{k-1}}}^{j} + (\mathbf{t}_{\mathbf{v}_{c_{k-1}}}^{j}) \Delta t \\ \mathbf{t}_{\mathbf{v}_{c_{k-1}}}^{j} + (\mathbf{t}_{\mathbf{v}_{c_{k-1}}}^{j}) \Delta t \\ \mathbf{t}_{\mathbf{v}_{c_{k-1}}}^{1} + (\mathbf{t}_{\mathbf{v}_{c_{k-1}}}^{j}) \Delta t \\ \mathbf{t}_{\mathbf{v}_{k-1}}^{1} - (\mathbf{v}_{t_{k-1}}) \Delta t \\ \vdots \\ \mathbf{t}_{\mathbf{x}_{k-1}}^{i} - (\mathbf{v}_{t_{k-1}}) \Delta t \end{bmatrix}$$

$$\mathbf{n}_{k} = \begin{bmatrix} \mathbf{t}_{\mathbf{y}_{k}}^{j} \\ \boldsymbol{\zeta}_{k} \end{bmatrix} = \begin{bmatrix} \mathbf{t}_{\mathbf{a}_{c}}^{j} \Delta t \\ \mathbf{a}_{t} \Delta t \end{bmatrix}$$
(33)

where ${}^{t}\mathbf{a_{c}}^{j}$ and $\mathbf{a_{t}}$ represent unknown linear accelerations that are assumed to have Gaussian distribution with zero mean. Moreover, let $\mathbf{n} \sim \mathcal{N}(\mathbf{0}, \mathbf{Q})$ be the noise vector that affects the state, while Δt is the differential of time and *k* the sample step. In this work, a Gaussian random process is used for propagating the velocity of the vehicle. The proposed scheme is independent of the kind of aircraft and therefore is not restricted by the use of an specific dynamic model.

An Extended Kalman Filter (EKF) propagates the system state $\hat{\mathbf{x}}$ over time as follows:

$$\hat{\mathbf{x}}_{k}^{-} = \mathbf{f}(\hat{\mathbf{x}}_{k-1}, \mathbf{0}) \tag{35}$$

The state covariance matrix *P* takes a step forward using [47]:

$$\mathbf{P}_{k}^{-} = \mathbf{A}_{k} \mathbf{P}_{k-1} \mathbf{A}_{k}^{T} + \mathbf{W}_{k} \mathbf{Q}_{k-1} \mathbf{W}_{k}^{T}$$
(36)

with

$$\mathbf{A}_{k} = \frac{\partial \mathbf{f}}{\partial \mathbf{x}}(\hat{\mathbf{x}}_{k-1}, \mathbf{0}) \qquad \mathbf{W}_{k} = \frac{\partial \mathbf{f}}{\partial \mathbf{n}}(\hat{\mathbf{x}}_{k-1}, \mathbf{0})$$
(37)

4.2. Measurement Updates

Assuming that, for the current sample step, a set of measurements is available, then the filter is updated with the Kalman update equations as follows [47]:

$$\hat{\mathbf{x}}_k = \hat{\mathbf{x}}_k^- + \mathbf{K}_k(\mathbf{z}_k - \mathbf{h}(\hat{\mathbf{x}}_k^-, \mathbf{0}))$$
(38)

$$\mathbf{P}_k = (\mathbf{I} - \mathbf{K}_k \mathbf{C}_k) \mathbf{P}_k^-$$
(39)

with

$$\mathbf{K}_{k} = \mathbf{P}_{k}^{-} \mathbf{C}_{k}^{T} (\mathbf{C}_{k} \mathbf{P}_{k}^{-} \mathbf{C}_{k}^{T} + \mathbf{V}_{k} \mathbf{R}_{k} \mathbf{V}_{k}^{T})^{-1}$$
(40)

and

$$\mathbf{C}_{k} = \frac{\partial \mathbf{h}}{\partial \mathbf{x}} (\hat{\mathbf{x}}_{k}^{-}, \mathbf{0}) \qquad \mathbf{V}_{k} = \frac{\partial \mathbf{h}}{\partial \mathbf{r}} (\hat{\mathbf{x}}_{k}^{-}, \mathbf{0})$$
(41)

where z is the vector of the current measurements, h is the vector of the current prediction measurements, and $r \sim \mathcal{N}(0, \mathbf{R})$ is the noise vector that affects the measurements.

4.2.1. Visual Updates (Landmarks)

When in the current sample step a set of visual measurements of the landmarks is available, the system is updated with this kind of measurements. In this case:

$$\mathbf{h} = \begin{bmatrix} {}^{1}\mathbf{h}_{\mathbf{c}}{}^{j} & \cdots & {}^{i}\mathbf{h}_{\mathbf{c}}{}^{j} \end{bmatrix}^{\mathrm{T}} \qquad \mathbf{z} = \begin{bmatrix} {}^{1}\mathbf{z}_{\mathbf{c}}{}^{j} & \cdots & {}^{i}\mathbf{z}_{\mathbf{c}}{}^{j} \end{bmatrix}^{\mathrm{T}}$$
(42)

4.2.2. Visual Updates (Target)

When in the current sample step a set of visual measurements of the target is available, the system is updated with this kind of measurements. In this case:

$$\mathbf{h} = \begin{bmatrix} \mathbf{t} \mathbf{h}_{\mathbf{c}}^{1} & \cdots & \mathbf{t} \mathbf{h}_{\mathbf{c}}^{j} \end{bmatrix}^{\mathrm{T}} \qquad \mathbf{z} = \begin{bmatrix} \mathbf{t} \mathbf{z}_{\mathbf{c}}^{1} & \cdots & \mathbf{t} \mathbf{z}_{\mathbf{c}}^{j} \end{bmatrix}^{\mathrm{T}}$$
(43)

4.2.3. Altitude Differential Updates

When in the current sample step a set of altitude differential measurements between pairs of UAVs is available, the system is updated with this kind of measurements. In this case:

$$\mathbf{h} = \begin{bmatrix} n h_a^j & \cdots \end{bmatrix}^{\mathrm{T}} \qquad \mathbf{z} = \begin{bmatrix} n z_a^j & \cdots \end{bmatrix}^{\mathrm{T}}$$
(44)

4.3. Map Features Initialization

The system state x is augmented with new map features using a cooperative strategy. The 3D relative position with respect to the target of the new map features is estimated using a pseudo-stereo system formed by the monocular cameras mounted on a pair of UAVs that observe common landmarks. In this case, when a new potential landmark is observed by two cameras, then it is initialized employing a linear triangulation.

The state of the new feature is computed using the a posteriori values obtained in the correction stage of the EKF. According to Equation (3), the following expression can be defined in homogeneous coordinates [52]:

$${}^{i}\gamma_{c}^{j}\begin{bmatrix}{}^{i}u_{c}^{j}\\ iv_{c}^{j}\\ 1\end{bmatrix} = \begin{bmatrix} \mathbf{T}_{c}^{j} & \mathbf{0}_{3\times 1} \end{bmatrix} \hat{\mathbf{E}}_{c}^{j}\begin{bmatrix} {}^{t}\mathbf{x}_{a}{}^{i}\\ 1\end{bmatrix}$$
(45)

where ${}^{i}\gamma_{c}^{j}$ is a scale factor. Additionally, it is defined:

$$\hat{\mathbf{E}}_{\mathbf{c}}^{j} = \begin{bmatrix} {}^{W}\mathbf{R}_{\mathbf{c}}{}^{j} & {}^{\mathbf{t}}\hat{\mathbf{x}}_{\mathbf{c}}{}^{j} \\ \mathbf{0}_{1\times3} & 1 \end{bmatrix}$$
$$\mathbf{T}_{\mathbf{c}}{}^{j} = \begin{bmatrix} \frac{f_{\mathbf{c}}^{j}}{d_{u}^{j}} & 0 & c_{u}^{j} + d_{ur}^{j} + d_{ut}^{j} \\ 0 & \frac{f_{\mathbf{c}}^{j}}{d_{v}^{j}} & c_{v}^{j} + d_{vr}^{j} + d_{vt}^{j} \\ 0 & 0 & 1 \end{bmatrix}$$
(46)

Using Equation (45), and considering the projection onto two any UAV-cameras, a linear system can be formed in order to estimate ${}^{t}\mathbf{x}_{a}^{i}$:

$$\mathbf{D}^{i} \cdot {}^{\mathbf{t}}\mathbf{x_{a}}^{i} = \mathbf{b}^{i} \qquad {}^{\mathbf{t}}\mathbf{x_{a}}^{i} = \mathbf{D}^{i^{\dagger}} \cdot \mathbf{b}^{i}$$

$$\tag{47}$$

where $\mathbf{D}^{i^{\dagger}}$ is the Moore Penrose right pseudo-inverse matrix of \mathbf{D}^{i} , and

$$\mathbf{D}^{i} = \begin{bmatrix} k_{31}^{j} \, {}^{i}u_{c}^{j} - k_{11}^{j} & k_{32}^{j} \, {}^{i}u_{c}^{j} - k_{12}^{j} & k_{33}^{j} \, {}^{i}u_{c}^{j} - k_{13}^{j} \\ k_{31}^{j} \, {}^{i}v_{c}^{j} - k_{21}^{j} & k_{32}^{j} \, {}^{i}v_{c}^{j} - k_{22}^{j} & k_{33}^{j} \, {}^{i}v_{c}^{j} - k_{23}^{j} \end{bmatrix}$$

$$\mathbf{b}^{i} = \begin{bmatrix} k_{24}^{j} - k_{34}^{j} \, {}^{i}u_{c}^{j} \\ k_{24}^{j} - k_{34}^{j} \, {}^{i}v_{c}^{j} \end{bmatrix}$$

$$(48)$$

with

$$\begin{bmatrix} \mathbf{T}_{\mathbf{c}}^{j} & \mathbf{0}_{3\times 1} \end{bmatrix} \hat{\mathbf{E}}_{\mathbf{c}}^{j} = \begin{bmatrix} k_{11}^{j} & k_{12}^{j} & k_{13}^{j} & k_{14}^{j} \\ k_{21}^{j} & k_{22}^{j} & k_{23}^{j} & k_{24}^{j} \\ k_{31}^{j} & k_{32}^{j} & k_{33}^{j} & k_{34}^{j} \end{bmatrix}$$
(49)

When a new landmark is initialized, the system state x is augmented by:

$$\mathbf{x} = \begin{bmatrix} \mathbf{t}_{\mathbf{x}_{c}}^{1} & \mathbf{t}_{\mathbf{v}_{c}}^{1} & \cdots & \mathbf{t}_{\mathbf{x}_{c}}^{j} & \mathbf{t}_{\mathbf{v}_{c}}^{j} & \mathbf{v}_{t} & \mathbf{t}_{\mathbf{x}_{a}}^{1} & \cdots & \mathbf{t}_{\mathbf{x}_{a}}^{i} & \mathbf{t}_{\mathbf{x}_{a}}^{new} \end{bmatrix}^{\mathrm{T}}$$
(50)

In addition, the new covariance matrix \mathbf{P}_{new} is computed by:

$$\mathbf{P}_{new} = \Delta \mathbf{J} \begin{bmatrix} \mathbf{P} & \mathbf{0} \\ \mathbf{0} & {}^{i}\mathbf{R}{}^{j} \end{bmatrix} = \Delta \mathbf{J}^{\mathrm{T}}$$
(51)

where $\Delta \mathbf{J}$ is the Jacobian for the initialization function, and ${}^{i}\mathbf{R}^{j}$ is the measurement noise covariance matrix for $({}^{i}u_{cr}^{j}, {}^{i}v_{c}^{j})$.

4.4. Map Management

It is well known that due to the nature of the Kalman Filter, in SLAM, the system state can always reach a size that will make it impossible to maintain a real-time performance for a given hardware configuration. In this sense, the present work is mainly intended to address the problem of navigation with respect to a target (local navigation). Therefore, features that are left behind by the movement of the cameras will be removed from the system state and covariance matrix. This strategy will prevent that the system state size reaches an unmanageable amount that seriously affects the computational performance.

5. Control System

This section presents the control scheme that allows for carrying out the flight formation of the UAVs with respect to the target. While the proposed control scheme is presented assuming that UAVs are quadcopters, the methodology can be easily extended for being used with any other aerial configurations.

5.1. Dynamic Model Flight Formation

The dynamic model used for representing the flight formation is based on the leader–follower scheme [64]. In this case, it is desired to maintain the *j*-th UAV to a distance ${}^{t}x_{q}^{j}$, ${}^{t}y_{q}^{j}$ (in the x - y plane) from the target (See Figure 4). In addition, it is also desired to maintain the *j*-th UAV at an altitude differential ${}^{t}z_{q}^{j}$ from the target (See Figure 4). Given the above considerations, the following expression can be defined:

$${}^{\mathbf{t}}\mathbf{x}_{\mathbf{q}}{}^{j} = \begin{bmatrix} {}^{t}x_{q}^{j} \\ {}^{t}y_{q}^{j} \\ {}^{t}z_{q}^{j} \end{bmatrix} = \mathbf{x}_{\mathbf{q}}{}^{j} - \mathbf{x}_{\mathbf{t}}$$
(52)



Figure 4. UAVs—target flight formation.

Differentiating twice Equation (52) with respect to time and using Equation (8), the dynamics of the formation ${}^{t}x_{q}{}^{j}$ can be obtained:

$${}^{\mathbf{t}}\ddot{\mathbf{x}}_{\mathbf{q}}{}^{j} = \ddot{\mathbf{x}}_{\mathbf{q}}^{j} - \ddot{\mathbf{x}}_{\mathbf{t}} = \frac{\mu^{j}}{m^{j}} \begin{bmatrix} \sin(\psi^{j})\sin(\phi^{j}) + \cos(\psi^{j})\sin(\theta^{j})\cos(\phi^{j}) \\ \sin(\psi^{j})\sin(\theta^{j})\cos(\phi^{j}) - \cos(\psi^{j})\sin(\phi^{j}) \\ \cos(\theta^{j})\cos(\phi^{j}) - \frac{g \cdot m^{j}}{u_{q}^{j} \cdot \mu^{j}} \end{bmatrix} u_{q}^{j} - \ddot{\mathbf{x}}_{\mathbf{t}}$$
(53)

5.2. Control Scheme

The control scheme is designed to allow that the flight formation converges to the desired values. The control system is divided into two subsystems (see Figure 5): (i) the flight formation control, which involves the translational dynamics of the quadcopter and the target; (ii) the rotational control. Because the reference signal of the rotational subsystem is obtained from the flight formation control subsystem, it is assumed that the dynamics of the former is faster than the dynamics of the latter.

A master–slave system is defined by:

$$\sum = \begin{cases} \mathbf{t} \ddot{\mathbf{x}}_{\mathbf{q}}^{\ j} = \ddot{\mathbf{x}}_{\mathbf{q}}^{\ j} - \ddot{\mathbf{x}}_{\mathbf{t}} \\ \ddot{\sigma}^{\ j} = \mathbf{g}^{\ j} + \mathbf{B}^{\ j} \tau^{\ j} \end{cases}$$
(54)

The overall control scheme is based on the Backstepping control technique [49], and each control-loop subsystem is designed by the so-called technique exact linearization by state feedback [49]. The state values required by the flight formation control subsystem are obtained from the estimator described in Section 4. The attitude of the *j*-th UAV is obtained assuming the availability of an on-board AHRS. To obtain the control laws using analysis in continuous time, it is assumed that the estimated values are passed through a zero-order hold (ZOH) (see Figure 5).



Figure 5. Control scheme diagram.

5.2.1. Flight Formation Subsystem Control

Since the estimated state of the relative position of the *j*-th UAV with respect to the target is defined with respect to the reference system C (see Sections 2 and 4), it is necessary to apply a transformation to the estimated state ${}^{t}\hat{\mathbf{x}}_{c}{}^{j}$ for obtaining ${}^{t}\hat{\mathbf{x}}_{q}{}^{j}$, which, in turn, is necessary to obtain the control laws. Therefore, the following equation is defined:

$${}^{t}\hat{\mathbf{x}}_{\mathbf{q}}^{\ j} = {}^{t}\hat{\mathbf{x}}_{\mathbf{c}}^{\ j} - {}^{\mathbf{q}}\mathbf{d}_{\mathbf{c}}^{\ j} \tag{55}$$

Let ${}^{\mathbf{q}}\mathbf{d}_{\mathbf{c}}{}^{j}$ be the translation vector (in meters) from the coordinate reference system Q to the coordinate reference system C. Note that ${}^{\mathbf{q}}\mathbf{d}_{\mathbf{c}}{}^{j}$ is assumed to be known and constant.

Firstly, a control law for the altitude differential ${}^{t}z_{q}^{j}$ of the flight formation subsystem is developed. Considering the dynamics of ${}^{t}z_{q}^{j}$ given in Equation (53), with control input u_{q}^{j} , and defining:

$$s_z^j = \dot{e}_z^j + \lambda_z^j e_z^j + \kappa_z^j \int_0^t e_z^j dt$$
(56)

where the error signal is $e_z^j = {}^t \hat{z}_q^j - {}^t z_{q_d}^j$ and ${}^t z_{q_d}^j$ is the desired value.

Deriving Equation (56) and substituting in Equation (53), it is obtained:

$$\dot{s}_{z}^{j} = \left(\frac{\mu^{j}}{m^{j}}\right) \left(\cos(\hat{\theta}^{j})\cos(\hat{\phi}^{j})\right) u_{q}^{j} - g + \lambda_{z}^{j}\dot{e}_{z}^{j} + \kappa_{z}^{j}e_{z}^{j} - \ddot{z}_{t} - {}^{t}\dot{z}_{qd}^{j}$$
(57)

The following control law is proposed:

$$u_{q}^{j} = \frac{m^{j}({}^{t}\ddot{z}_{q_{d}}^{j} + \ddot{z}_{t} - \lambda_{z}^{j}\dot{e}_{z}^{j} - \kappa_{z}^{j}e_{z}^{j} - k_{z}^{j}s_{z}^{j} + g)}{\mu^{j}(\cos(\hat{\theta}^{j})\cos(\hat{\phi}^{j}))}$$
(58)

where k_z^j , λ_z^j , and κ_z^j are positive gains assuming that $\hat{\theta}^j$, $\hat{\phi}^j \in [-\frac{\pi}{2}, \frac{\pi}{2}]$.

Now, consider the longitudinal and lateral differential ${}^{t}x_{q}^{j}$ and ${}^{t}y_{q}^{j}$ of the flight formation subsystem. By substituting Equation (58) in Equation (53), then, for the two first states, it can be obtained:

$$\begin{bmatrix} t \ddot{x}_{q}^{j} \\ t \ddot{y}_{q}^{j} \end{bmatrix} = u_{z}^{j} \begin{bmatrix} \sin(\hat{\psi}^{j}) & \cos(\hat{\psi}^{j}) \\ -\cos(\hat{\psi}^{j}) & \sin(\hat{\psi}^{j}) \end{bmatrix} \begin{bmatrix} \frac{\tan(\phi^{j})}{\cos(\hat{\theta}^{j})} \\ \tan(\theta^{j}) \end{bmatrix} - \begin{bmatrix} \ddot{x}_{t} \\ \ddot{y}_{t} \end{bmatrix}$$
(59)

Taking ϕ^{j} and θ^{j} as control inputs in Equation (59), and defining:

$$s_x^j = \dot{e}_x^j + \lambda_x^j e_x^j + \kappa_x^j \int_0^t e_x^j dt$$

$$s_y^j = \dot{e}_y^j + \lambda_y^j e_y^j + \kappa_y^j \int_0^t e_y^j dt$$
(60)

with error signals $e_x^j = {}^t \hat{x}_q^j - {}^t x_{q_d}^j$ and $e_y^j = {}^t \hat{y}_q^j - {}^t y_{q_d}^j$, being ${}^t x_{q_d}^j$ and ${}^t y_{q_d}^j$ the desired values, it is possible to derive Equation (60) and substitute in Equation (59) in order to obtain:

$$\begin{bmatrix} \dot{s}_{x}^{j} \\ \dot{s}_{y}^{j} \end{bmatrix} = u_{z}^{j} \begin{bmatrix} \sin(\hat{\psi}^{j}) & \cos(\hat{\psi}^{j}) \\ -\cos(\hat{\psi}^{j}) & \sin(\hat{\psi}^{j}) \end{bmatrix} \begin{bmatrix} \frac{\tan(\phi^{j})}{\cos(\hat{\theta}^{j})} \\ \tan(\theta^{j}) \end{bmatrix} + \begin{bmatrix} \lambda_{x}^{j} \dot{e}_{x}^{j} + \kappa_{x}^{j} \dot{e}_{x}^{j} - t_{x}^{j} - \ddot{x}_{t} \\ \lambda_{y}^{j} \dot{e}_{y}^{j} + \kappa_{y}^{j} e_{y}^{j} - t_{y}^{j} \dot{q}_{d} - \ddot{y}_{t} \end{bmatrix}$$
(61)

The following control laws are proposed:

$$\begin{bmatrix} \theta^{j} \\ \phi^{j} \end{bmatrix} = \arctan\left(\frac{1}{u_{z}^{j}} \begin{bmatrix} \cos(\hat{\psi}^{j}) & \sin(\hat{\psi}^{j}) \\ \sin(\hat{\psi}^{j})\cos(\hat{\theta}^{j}) & -\cos(\hat{\psi}^{j})\cos(\hat{\theta}^{j}) \end{bmatrix} \begin{bmatrix} u_{x}^{j} \\ u_{y}^{j} \end{bmatrix}\right)$$
(62)

with

$$u_{x}^{j} = {}^{t} \ddot{x}_{q_{d}}^{j} + \ddot{x}_{t} - \lambda_{x}^{j} \dot{e}_{x}^{j} - \kappa_{x}^{j} e_{x}^{j} - k_{x}^{j} s_{x}^{j}$$

$$u_{y}^{j} = {}^{t} \ddot{y}_{q_{d}}^{j} + \ddot{y}_{t} - \lambda_{y}^{j} \dot{e}_{y}^{j} - \kappa_{y}^{j} e_{y}^{j} - k_{y}^{j} s_{y}^{j}$$

$$u_{z}^{j} = {}^{t} \ddot{z}_{q_{d}}^{j} + \ddot{z}_{t} - \lambda_{z}^{j} \dot{e}_{z}^{j} - \kappa_{z}^{j} e_{z}^{j} - k_{z}^{j} s_{z}^{j} + g$$
(63)

and where k_x^j , λ_x^j , κ_x^j , k_y^j , λ_y^j and κ_y^j are positive gains.

For the proposed control laws in Equations (58) and (62), it is necessary to know the accelerations of the target, which are estimated by deriving the estimated velocities of the target obtained from the EKF by using the following differentiator with the super-twisting algorithm [65] (see Figure 5):

$$\ddot{\mathbf{x}}_{\mathbf{t}} = -\mathbf{K}_{\mathbf{1}} \left| \dot{\mathbf{x}}_{\mathbf{t}} - \hat{\mathbf{v}}_{\mathbf{t}} \right|^{\frac{1}{2}} \operatorname{sign}(\dot{\mathbf{x}}_{\mathbf{t}} - \hat{\mathbf{v}}_{\mathbf{t}}) + \hat{\omega}_{t} \dot{\hat{\omega}}_{t} = -\mathbf{K}_{\mathbf{2}} \operatorname{sign}(\dot{\mathbf{x}}_{\mathbf{t}} - \hat{\mathbf{v}}_{\mathbf{t}})$$
(64)

where $\hat{\mathbf{x}}_t$ is the velocity of the target estimated by the differentiator, $\hat{\mathbf{v}}_t$ is the velocity of the target estimated by the EKF, \mathbf{K}_1 and \mathbf{K}_2 are positive definite diagonal matrices of adequate dimensions.

To prove the stability of the flight formation subsystem, the following theorem is used:

Theorem 1. Consider the dynamic in Equations (57) and (61) in closed loop with Equations (58) and (62); then, the origin $s_x^j = s_y^j = s_z^j = 0$ of Equations (56) and (60) is asymptotically stable.

Proof. Consider the following Lyapunov candidate function:

$$V_t = \frac{1}{2} \mathbf{s_t}^{j^{\mathrm{T}}} \mathbf{s_t}$$
(65)

with $\mathbf{s_t}^j = \begin{bmatrix} s_x^j & s_y^j & s_z^j \end{bmatrix}^{\mathrm{T}}$. By deriving:

$$\dot{V}_{t} = \mathbf{s_{t}}^{j^{T}} \dot{\mathbf{s}}_{t}^{j} = \mathbf{s_{t}}^{j^{T}} \begin{bmatrix} u_{z}^{j} \sin(\hat{\psi}^{j}) & u_{z}^{j} \cos(\hat{\psi}^{j}) & 0 \\ -u_{z}^{j} \cos(\hat{\psi}^{j}) & u_{z}^{j} \sin(\hat{\psi}^{j}) & 0 \\ 0 & 0 & \left(\frac{\mu^{j}}{m^{j}}\right) \cos(\hat{\phi}^{j}) \cos(\hat{\theta}^{j}) \end{bmatrix} \begin{bmatrix} \frac{\tan(\phi^{j})}{\cos(\hat{\theta}^{j})} \\ \tan(\theta^{j}) \\ u_{q}^{j} \end{bmatrix} + \mathbf{s_{t}}^{j^{T}} \begin{bmatrix} \lambda_{x}^{j} e_{x}^{j} + \kappa_{x}^{j} e_{x}^{j} - t x_{q_{d}}^{j} - \ddot{x}_{t} \\ \lambda_{y}^{j} e_{y}^{j} + \kappa_{y}^{j} e_{y}^{j} - t y_{q_{d}}^{j} - \ddot{y}_{t} \\ -g + \lambda_{z}^{j} e_{z}^{j} + \kappa_{z}^{j} e_{z}^{j} - t z_{q_{d}}^{j} - \ddot{z}_{t} \end{bmatrix}$$
(66)

Substituting Equations (58) and (62) in Equation (66), it is obtained:

$$\dot{V}_t = -\mathbf{s_t}^{j^{\mathrm{T}}} \mathbf{K_t}^j \mathbf{s_t}^j \le 0 \tag{67}$$

with

$$\mathbf{K}_{\mathbf{t}}^{j} = \begin{bmatrix} k_{x}^{j} & 0 & 0\\ 0 & k_{y}^{j} & 0\\ 0 & 0 & k_{z}^{j} \end{bmatrix}$$
(68)

Therefore, the origin $\mathbf{s}_t^j = \mathbf{0}$ of Equations (56) and (60) is asymptotically stable. \Box

5.2.2. Rotational Subsystem Control

For the rotational subsystem, with dynamics characterized by Equation (9), and control input τ^{j} , it can be defined:

$$\mathbf{s}_{\mathbf{r}}^{j} = \dot{\mathbf{e}}_{\mathbf{r}}^{j} + \lambda_{\mathbf{r}}^{j} \mathbf{e}_{\mathbf{r}}^{j} + \kappa_{\mathbf{r}}^{j} \int_{0}^{t} \mathbf{e}_{\mathbf{r}}^{j} dt$$
(69)

where the error signal is $\mathbf{e}_{\mathbf{r}}^{j} = \sigma^{j} - \sigma_{d}^{j}$ and σ_{d}^{j} is the desired value. Deriving Equation (69) and substituting in Equation (9), it is obtained:

$$\dot{\mathbf{s}}_{\mathbf{r}} = \mathbf{g}^{j} + \mathbf{B}^{j} \boldsymbol{\tau}^{j} + \lambda_{\mathbf{r}}^{j} \dot{\mathbf{e}}_{\mathbf{r}}^{j} + \kappa_{\mathbf{r}}^{j} \mathbf{e}_{\mathbf{r}}^{j} - \sigma_{d}^{j}$$
(70)

The following control law is proposed:

$$\boldsymbol{\tau}^{j} = \mathbf{B}^{j-1} \left(-\mathbf{g}^{j} + \boldsymbol{\sigma}_{d}^{j} - \boldsymbol{\lambda}_{r}^{j} \dot{\mathbf{e}}_{\mathbf{r}}^{j} - \boldsymbol{\kappa}_{r}^{j} \mathbf{e}_{\mathbf{r}}^{j} - \mathbf{K}_{\mathbf{r}}^{j} \mathbf{s}_{\mathbf{r}}^{j} \right)$$
(71)

where λ_r^{j} , κ_r^{j} , and \mathbf{K}_r are positive definite diagonal matrices of adequate dimensions. It is straightforward to demonstrate that $\mathbf{B}^{j^{-1}}$ exists.

To prove the stability of the rotational subsystem, the following theorem is used:

Theorem 2. Consider the dynamic in Equation (70) in closed loop with Equation (71); then the origin $\mathbf{s_r}^j = \mathbf{0}$ of Equation (69) is asymptotically stable.

Proof. Consider the following Lyapunov candidate function:

$$V_r = \frac{1}{2} \mathbf{s_r}^{j^{\mathrm{T}}} \mathbf{s_r}$$
(72)

By deriving:

$$\dot{V}_{r} = \mathbf{s_{r}}^{j^{\mathrm{T}}} \dot{\mathbf{s}}_{r}^{j} = \mathbf{s_{r}}^{j^{\mathrm{T}}} \left(\mathbf{g}^{j} + \mathbf{B}^{j} \boldsymbol{\tau}^{j} + \lambda_{r}^{j} \dot{\mathbf{e}}_{r}^{j} + \kappa_{r}^{j} \mathbf{e}_{r}^{j} - \sigma_{d}^{j} \right)$$
(73)

Substituting Equation (71) in Equation (73), it is obtained:

$$\dot{V}_{\mathbf{r}} = -\mathbf{s_r}^{j^{\mathrm{T}}} \mathbf{K_r}^{j} \mathbf{s_r}^{j} \le 0$$
(74)

Therefore, the origin $\mathbf{s_r}^j = \mathbf{0}$ of Equation (69) is asymptotically stable. \Box

5.2.3. Closed Loop Stability

In order to prove the closed-loop stability of the whole system in Equation (54), the following theorem is used:

Theorem 3. Given the system in Equation (54), in closed loop with the control laws in Equations (58) and (71). Moreover, if θ_d^j and ϕ_d^j are chosen according to Equation (62) as control laws for the longitudinal and lateral differential of the flight formation subsystem, then the origin $\mathbf{s_t}^j = \mathbf{s_r}^j = \mathbf{0}$ is asymptotically stable.

Proof. Consider the following Lyapunov candidate function:

$$V_T = V_t + V_r \tag{75}$$

with derivative

$$\dot{V}_T = \dot{V}_t + \dot{V}_r = \mathbf{s_t}^{j^1} \dot{\mathbf{s}}_t^j + \mathbf{s_r}^{j^1} \dot{\mathbf{s}}_r^j$$
(76)

from Equations (67) and (74), it can be seen that

$$\dot{V}_T = -\mathbf{s_t}^{j^T} \mathbf{K_t} \mathbf{s_t} - \mathbf{s_r}^{j^T} \mathbf{K_r} \mathbf{s_r} \le 0$$
(77)

Therefore, the origin $\mathbf{s}_t^j = \mathbf{s}_r^j = \mathbf{0}$ of the closed loop system is Globally Asymptotically Stable (GAS). \Box

6. Simulation Results

6.1. Simulation Setup

In this section, the proposed cooperative visual-SLAM system for UAV-based target tracking is validated through computer simulations. A Matlab^(R) implementation of the proposed scheme was used for this purpose. To this aim, a simulation environment has been developed. The environment is composed of 3D landmarks, randomly distributed over the ground (See Figure 6). To execute the tests, two quadcopters (Quad 1 and Quad 2), equipped with the set of sensors required by the proposed method are simulated. In simulations, it is assumed that there exist enough landmarks in the environment that allow for being observed in common by the cameras of the UAVs, at least a subset of them.

The measurements from the sensors are emulated to be taken with a frequency of 10 Hz. To emulate the system uncertainty, the following Gaussian noise is added to measurements: Gaussian noise with $\sigma_c = 4$ pixels is added to the measurements given by the cameras. Gaussian noise with $\sigma_a = 25$ cm is added to the measurements of altitude differential between two quadcopters obtained by the altimeters. It is important to note that the noise considered for emulating monocular measurements is bigger than the typical magnitude of the noise of real monocular measurement. In this way, the errors in camera orientation are considered caused by the gimbal, in addition to the imperfection of the sensor.



In simulations, the target was moved along a predefined trajectory (see Figure 6).

Figure 6. Actual trajectories followed by the target, Quad 1 and Quad 2.

The parameter values used for the quadcopters and the cameras are shown in Table 2. The quadcopters' parameters are like those presented in [66]. The camera parameters are like those presented in [17].

Parameter	Value	Parameter	Value
m^j	0.468	lj	0.225
μ^{j}	$2.98 imes10^{-6}$	8	9.81
J_p^j	$3.357 imes10^{-5}$	f^j/d_u^j	262.92
d^{j}	$1.14 imes 10^{-7}$	f^j/d_v^j	261.66
I_x^j, I_y^j	$4.856 imes10^{-3}$	c_u^j	359.51
I_z^j	$8.801 imes 10^{-3}$	c_v^j	239.50

Table 2. Quadcopter and camera parameters.

Figure 6 shows the trajectories of the three elements composing the flight formation. To highlight them, the position of the elements of the formation is indicated at different instants of time.

6.2. Convergence and Stability Tests

In a series of tests, the convergence and stability of the proposed cooperative visual-SLAM system, in open-loop, were evaluated under the three different observability conditions described in Section 3.2. In this case, it is assumed that there is a control system able to maintain the aerial robots flying in formation with respect to the target.

Two different kinds of tests were performed:

- (a) For the first test, the robustness of the cooperative visual-SLAM system with respect to errors in the initialization of map features is tested. In this case, the initial conditions of ${}^{t}\hat{\mathbf{x}_{c}}^{1}$, ${}^{t}\hat{\mathbf{x}_{c}}^{2}$, ${}^{t}\hat{\mathbf{v}_{c}}^{1}$, ${}^{t}\hat{\mathbf{v}_{c}}^{2}$ and $\hat{\mathbf{v}}_{t}$ are assumed to be known exactly, but each *i* map feature is forced to be initialized into the system state with a determined error position. Three different conditions of initial error are considered: $\sigma_{i} = \{0.5, 1.5, 2.25\}$ meters.
- (b) For the second test, additional to the map features initial errors, the initial conditions of ${}^{t}\hat{\mathbf{x}}_{c}^{1}$, ${}^{t}\hat{\mathbf{x}}_{c}^{2}$, ${}^{t}\hat{\mathbf{v}}_{c}^{1}$, ${}^{t}\hat{\mathbf{v}}_{c}^{2}$, and $\hat{\mathbf{v}}_{t}$ are not known exactly and have a considerable error with respect to the real values. In this case, the robustness of the proposed estimation system is evaluated in function to the errors in the initial conditions of all the state variables.

Figure 7 and Figure 8 show respectively the results for tests (a) and (b). In this case, the estimated relative position of the Quad 1 with respect to the target is plotted for each reference axis (row plots). Note that, for the sake of clarity, only the estimated values for Quad 1 are presented. The results for Quad 2 are quite similar to those presented for Quad 1. In these figures, each column of plots shows the results obtained from an observability case.

In both figures, it can be observed that both the observability property and the initial conditions play a preponderant role in the convergence and stability of the EKF-SLAM. For several applications, at least the initial position of the UAVs is assumed to be known. However, in SLAM, the position of the map features must be estimated online. The above outcome confirms the importance of using good feature initialization techniques in visual-SLAM. Of course, as it can be expected, the performance of the EKF-SLAM is considerably better when the system is observable. Note that, for the observability case 3, and despite a noticeable error drift, the filter still exhibits certain stability for the worst case of features initialization, when the initial conditions of the UAVs are known.



Figure 7. The estimated relative position of the Quad 1 with respect to the target obtained by the Test (a).



Figure 8. The estimated relative position of the Quad 1 with respect to the target obtained by the Test (b).

6.3. Comparative Study

Using the same simulation setup, for this series of tests, the performance of the proposed system, for estimating the relative position of the UAVs with respect to the moving target, was evaluated through a comparative study. Table 3 summarizes the characteristics of five UAV-based target-tracking methods used in this comparative study.

23 of 33

Method	System Configuration	Parametrization	Visual Measurements	Other Measurements
Proposed	Multi-UAV-SLAM	Target-centric	Monocular	Altitude differential
(1)	Single-UAV-SLAM	Target-centric	Monocular	-
(2)	Multi-UAV-SLAM	Robot-centric	Monocular	Altitude differential
(3)	Single-UAV	Robot-centric	Stereo-vision	-
(4)	Multi-UAV	Robot-centric	Pseudo-stereo	GPS
(5)	Single-UAV	Robot-centric	Monocular	Range

Table 3. Characteristics of the methods used in the comparative study.

There are some remarks about the methods used in the comparison. Method (1) represents a purely standard monocular-SLAM approach. Features initialization is based on [67]. Because the metric scale cannot be retrieved using only monocular vision, it is assumed that the positions of landmarks seen in the first frame are perfectly known. Method (2) is similar to the proposed previous method, but the system is robot-centric parametrized, instead of target-centric. Method (3) represents an standard visual-stereo tracking approach. In this case, the moving target position is directly obtained by a stereo system, with a baseline of 15 cm. The method (4) is based on the approach presented in [68]. In this case, UAVs are equipped with GPS, and the moving target position is estimated through the pseudo-stereo vision system composed of the monocular cameras of each UAV. Gaussian noise with $\sigma_{o} = 20$ cm is added to GPS measurements. Method (5) estimates the target position using monocular and range measurements to the target. In this case, it is assumed some cooperative-target scheme for obtaining range measurements. Technology for obtaining such kind of range measurements is presented in [69]. Gaussian noise with $\sigma_r = 20$ cm is added to the range measurements. Note that methods (3), (4), and (5) are not SLAM methods. Methods (3) and (5) provide only the relative estimation of the target position with respect to the UAV. Method (4), due to the GPS, provides also global position estimations of the target and UAVs.

In case of SLAM methods and to carry out a more realistic evaluation, the data association problem was also accounted for. To emulate the failures of the visual data association process, 5% of the total number of visual correspondences are forced to be outliers in a random manner. Table 4 shows the number of outliers introduced into the simulation due to the data association problem. Outliers for the visual data association in each camera as well as outliers for the cooperative visual data association are considered.

Method	Visual Outliers (Quad 1)	Visual Outliers (Quad 2)	Visual Outliers (Cooperative)
Proposed	3798	3798	2338
(1)	3820	-	-
(2)	3798	3798	2338

Table 4. Number of outliers introduced into the SLAM methods.

Figure 9 shows the results obtained from each method for estimating the relative position of Quad 1 with respect to the target. For the sake of clarity, the results are shown in two columns of plots. Each row of plots represents a reference axis.



Figure 9. The estimated relative position of the Quad 1 with respect to the target obtained from the six systems.

Table 5 summarizes the Mean Squared Error (MSE) for the estimated relative position of Quad 1 with respect to the target in the three axes.

Table 5. Total Mean Squared Error of the estimated relative position of Quad 1 with respect to the target.

Method	MSEX (m)	MSEY (m)	MSEZ (m)
Proposed	0.0029	0.0032	0.0512
(1)	0.0975	0.1053	6.0702
(2)	0.0083	0.0090	0.5238
(3)	0.0069	0.0048	0.0937
(4)	0.2829	0.3539	23.4294
(5)	0.0045	0.0036	0.0230

Regarding the performance of SLAM methods for estimating the features map, Table 6 shows the total (sum of all) Mean Squared Errors for the estimated position of landmarks, while Table 7 shows the total Mean Squared Errors for the initial estimated position of the landmarks.

Method	MSEX (m)	MSEY (m)	MSEZ (m)
Proposed	0.0969	0.0687	0.1991
(1)	6.1459	6.2934	5.9953
(2)	0.4231	0.3471	0.7477

Table 6. Total Mean Squared Error of the estimated position of landmarks.

Table 7. Total Mean Squared Error of the estimated initial position of landmarks.

Method	MSEX (m)	MSEY (m)	MSEZ (m)
Proposed	0.9993	0.8787	1.2064
(1)	4.3668	4.4917	3.8829
(2)	1.4241	1.2655	1.7639

According to the above results, the relative position of the UAV with respect to the moving target was recovered fairly well with the following methods: authors' proposed Target-Centric Multi-UAV SLAM approach, (2) Robot-Centric Multi-UAV SLAM, (4) Multi-UAV Pseudo-stereo, and (5) Single-UAV monocular-range.

It is important to note that method (4) relies on GPS and thus it is not suitable for GPS-denied environments. Method (5) relies on range measurements to the target which can be often difficult to obtain, or it requires some cooperative target scheme which is not even possible to accomplish for several kinds of applications. Regarding the other approaches, method (1) does not have direct three-dimensional information of the target (only monocular measurement) and thus the estimation is not good. In the case of method (3), it is well known that, with a fixed stereo system, the quality of measurements considerably degenerates as the distance with respect to the target increases.

Thus far, the above results suggest that both the proposed approach and the method (2) (both SLAM systems) exhibit good performance in challenging conditions: (i) GPS-denied environments, (ii) non-cooperative target, and (iii) long distance to the target. The only difference between these two methods is the parametrization of the system (Target-centric vs. Robot-centric). To investigate if there is an advantage in the use of one parametrization over the other, another test was carried out.

In this case, using the same simulation setup, the robustness of the methods was tested against the loss of visual contact to the moving target by one UAV during some period. Figure 10 shows the estimated relative positions of Quad 1 with respect to the target obtained with both systems. Note that, during the seconds 45 to 75 of simulation, there are no monocular measurements of the target done by Quad 1. Only the estimated values of Quad 1 are presented, but the results of Quad 2 are very similar.

Given the above results, it can be seen that, contrary to the method (2), the proposed system is robust to the loss of visual contact of the target by one of the Quads. This is a clear advantage of the proposed Target-centric parameterization.



Figure 10. Estimated relative position of Quad 1 with respect to the target with a period of loss of visual contact of the target by Quad 1.

6.4. Control Simulations' Results

A set of simulations was also carried out to test the whole proposed system in an estimation-control closed-loop manner. In this case, to maintain a stable flight formation with respect to the moving target, the estimates obtained from the proposed visual-based estimation system are used as feedback to the control scheme described in Section 5).

In this test, the values of vectors ${}^{t}\mathbf{x}_{\mathbf{q}_{d}}^{i}$ that define the flight formation are:

For Quad 1:
$${}^{t}\mathbf{x}_{\mathbf{q}_{d}}^{1} = \begin{bmatrix} -1.4142 - 0.35\sin(t \cdot 0.1), & 1.4142 + 0.35\sin(t \cdot 0.1), & 12 \end{bmatrix}^{T}$$
.
For Quad 2: ${}^{t}\mathbf{x}_{\mathbf{q}_{d}}^{2} = \begin{bmatrix} 2 + 0.5\sin(t \cdot 0.1), & 0, & 14 \end{bmatrix}^{T}$.

In addition, in this test, a period of visual contact loss of target by Quad 1 is simulated (seconds 35 to 50). A period of a visual contact loss of target by both Quads is simulated (seconds 90 to 95) as well.

Figure 11 shows the evolution of the error with respect to the desired values ${}^{t}x_{qd}^{i}$. In all the cases, note that the errors are bounded after an initial transient period.



Figure 11. Errors in ${}^{t}\mathbf{x}_{\mathbf{q}_{d}}^{i}$ during the flight trajectories.

Figure 12 shows the real and estimated relative position of Quad 1 (left column) and Quad 2 (right column) with respect to the target. Table 8 summarizes the Mean Squared Error (MSE), in each axis, for the estimated relative position of Quad 1 and Quad 2 with respect to the target.

Table 8. Total Mean Squared Error in the estimated relative position of Quad 1 and Quad 2 with respect to the target.

	MSEX (m)	MSEY (m)	MSEZ (m)
Quad 1	0.0137	0.0080	0.0738
Quad 2	0.0134	0.0068	0.0848

According to the above results, it can be seen that the control system was able to maintain a stable flight formation along with all the trajectory with respect to the target, using the proposed visual-based estimation system as feedback.



Figure 12. The estimated relative position of Quad 1 and Quad 2 with respect to the target.

7. Conclusions

This research presents a cooperative visual-based SLAM system that allows a team of aerial robots autonomously following a non-cooperative target moving freely in a GPS-denied environment. The relative position of each UAV with respect to the target is estimated by fusing monocular measurements of the target and landmarks using the standard EKF-based SLAM methodology.

The observability property of the system was investigated using an extensive nonlinear observability analysis. In this case, new theoretical results were presented and the observability conditions were derived. To improve the observability properties of the system, and thus its performance, the use of a Target-centric SLAM configuration is proposed. In addition, measurements of altitude differential between pairs of UAVs, obtained from altimeters, are integrated into the system for the same purpose. Additionally to the proposed estimation system, a control scheme was proposed allowing the control of UAVs flight formation with respect to the moving target. The stability of control laws is proved using the Lyapunov theory. An extensive set of computer simulations was performed in order to validate the proposed scheme. According to the simulation results, the proposed system shows an excellent and robust performance to estimate the relative position of each UAV with respect to target. The results also suggest that the proposed approach has better performance when it is compared with other related methods. Moreover, with the proposed control laws, the contributed SLAM system demonstrates good performance in closed-loop. On the other hand, although computer simulations are useful for evaluating the full statistical consistency of the methods, they can still neglect important practical issues that appear when the methods are applied in real scenarios. Accordingly,

it is important to note that future work will be focused on developing experiments with real data in order to fully evaluate the applicability of the proposed approach.

Author Contributions: Conceptualization, R.M. and A.G.; methodology, S.U and R.M.; software, J.-C.T. and S.U.; validation, J.-C.T., S.U., and A.G.; investigation, S.U. and R.M.; resources, J.-C.T. and S.U.; writing—original draft preparation, J.-C.T. and R.M.; writing—review and editing, R.M. and A.G.; supervision, R.M. and A.G.; funding acquisition, A.G. All authors have read and agreed to the published version of the manuscript.

Funding: This research has been funded by Project DPI2016-78957-R, Spanish Ministry of Economy, Industry and Competitiveness.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Xu, Z.; Douillard, B.; Morton, P.; Vlaskine, V. Towards Collaborative Multi-MAV-UGV Teams for Target Tracking. In 2012 Robotics: Science and Systems workshop Integration of Perception with Control and Navigation for Resource-Limited, Highly Dynamic, Autonomous Systems; Springer: Berlin, Germany, 2012.
- 2. Michael, N.; Shen, S.; Mohta, K. Collaborative mapping of an earthquake-damaged building via ground and aerial robots. *J. Field Robot.* **2012**, *29*, 832–841. [CrossRef]
- 3. Pappas, H.; Tanner, G.; Kumar, V. Leader-to-formation stability. IEEE Trans. Robot. Autom. 2004, 20, 443–455.
- 4. Zhu, Z.; Roumeliotis, S.; Hesch, H.; Park, H.; Venable, D. Architecture for Asymmetric Collaborative Navigation. In Proceedings of the 2012 IEEE/ION Position, Location and Navigation Symposium, PLANS 2012, Myrtle Beach, SC, USA, 23–26 April 2012; pp. 777–782.
- Hu, H.; Wei, N. A study of GPS jamming and anti-jamming. In Proceedings of the 2nd International Conference on Power Electronics and Intelligent Transportation System (PEITS), Shenzhen, China, 19–20 December 2009; Volume 1, pp. 388–391.
- 6. Bachrach, S.; Prentice, R.H.; Roy, N. RANGE-Robust autonomous navigation in GPS-denied environments. *J. Field Robot.* **2011**, *5*, 644–666. [CrossRef]
- 7. Meguro, J.I.; Murata, T.; Takiguchi, J.I.; Amano, Y.; Hashizume, T. GPS multipath mitigation for urban area using omnidirectional infrared camera. *IEEE Trans. Intell. Transp. Syst.* **2009**, *10*, 22–30. [CrossRef]
- De la Cruz, C.F.; Carelli, R.; Bastos, T. Navegación Autónoma asistida basada en SLAM para una silla de ruedas robotizada en entornos restringidos. *Revista Iberoamericana de Automática e Informática Ind.* 2011, 8, 81–92.
- 9. Andert, F.; Lorenz, S.; Mejias, L.; Bratanov, D. Radar-Aided Optical Navigation for Long and Large-Scale Flights over Unknown and Non-Flat Terrain. In Proceedings of the International Conference on Unmanned Aircraft Systems (ICUAS), Arlington, VA, USA, 7–10 June 2016.
- 10. Vázquez-Martín, R.; Nuñez, P.; Bandera, A.; Sandoval, F. Curvature Based Environment Description for Robot Navigation using Laser Range Sensors. *Sensors* **2009**, *9*, 5894–5918. [CrossRef]
- 11. Rao, K.S.; Nikitin, P.V.; Lam, S.F. Antenna design for UHF RFID tags: A review and a practical application. *IEEE Trans. Antennas Propag.* **2005**, *53*, 3870–3876. [CrossRef]
- 12. Opromolla, R.; Fasano, G.; Rufino, G.; Grassi, M. Hardware in the Loop Performance Assessment of LIDAR-Based Spacecraft Pose Determination. *Sensors* **2017**, *17*, 2197. [CrossRef]
- 13. Pirshayan, A.; Seyedarabi, H.; Haghipour, S. Cooperative Machine-Vision-Based Tracking using Multiple Unmanned Aerial Vehicles. *Adv. Comput. Sci. Int. J.* **2014**, *3*, 118–123.
- 14. Ophoff, T.; Van Beeck, K.; Goedemé, T. Exploring RGB+Depth Fusion for Real-Time Object Detection. *Sensors* **2019**, *19*, 866. [CrossRef]
- 15. Dib, A.; Zaidi, N.; Siguerdidjane, H. Robust Control and Visual Servoing of an UAV. In Proceedings of the 17th World Congress The International Federation of Automatic Control, Seoul, Korea, 6–11 July 2008.
- 16. Jabbari, H.; Oriolo, G.; Hossein, B. An Adaptive Scheme for Image-Based Visual Servoing of an Underactuated UAV. *Int. J. Robot. Autom.* **2013**, *29*, 92–104.
- 17. Vasquéz-Beltrán, A.M.; Rodríguez-Cortés, H. Seguimiento de una referencia visual en un plano con un cuatrirotor. In Proceedings of the Memorias del Congreso Nacional de Control Automático AMCA, Cuernavaca, Morelos, México, 14–16 October 2015.
- 18. Metni, N.; Hamel, T. Visual Tracking Control of Aerial Robotic Systems with Adaptive Depth Estimation. *Int. J. Control Autom. Syst.* **2007**, *5*, 51–60.

- 19. Rubio, J.d.J.; Zamudio, Z.; Meda, J.A.; Moreno, M.A. Experimental Vision Regulation of a Quadrotor. *IEEE Latin Am. Trans.* **2015**, *13*, 2514–2523. [CrossRef]
- 20. Dias, A.; Almeida, J.; Silva, E.; Lima, P. Uncertainty based multirobot cooperative triangulation. In *RoboCup Symposium*; Lecture Notes in Artificial Intelligence (LNAI); Springer: Cham, Switzerland, 2014.
- Wong, E.M.; Bourgault, F.; Furukawa, T. Multi-vehicle Bayesian search for multiple lost targets. In Proceedings of the 2005 IEEE International Conference on Robotics and Automation, Barcelona, Spain, 18–22 April 2005; pp. 3169–3174.
- Morbidi, F.; Mariottini, G.L. On active target tracking and cooperative localization for multiple aerial vehicles. In Proceedings of the2011 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), San Francisco, CA, USA, 25–30 September 2011; pp. 2229–2234.
- Dias, A.; Capitan, J.; Merino, L.; Almeida, J.; Lima, P.; Silva, E. Decentralized Target Tracking based on Multi-Robot Cooperative Triangulation. In Proceedings of the IEEE Robotics and Automation Society's Flagship Conference, Seattle, WA, USA, 26–30 May 2015.
- Dutta, R.; Sun, L.; Kothari, M.; Sharma, R.; Pack, D. A cooperative formation control strategy maintaining connectivity of a multi-agent system. In Proceedings of the 2014 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS 2014), Chicago, IL, USA, 14–18 September 2014; pp. 1189–1194.
- Hafez, A.T.; Marasco, A.J.; Givigi, S.N.; Beaulieu, A.; Rabbath, C.A. Encirclement of multiple targets using model predictive control. In Proceedings of the 2013 IEEE American Control Conference (ACC), Washington, DC, USA, 17–19 June 2013; pp. 3147–3152.
- 26. Kothari, M.; Sharma, R.; Postlethwaite, I.; Beard, R.W.; Pack, D. Cooperative target-capturing with incomplete target information. *J. Intell. Robot. Syst.* **2013**, *72*, 373–384. [CrossRef]
- Marasco, A.J.; Givigi, S.N.; Rabbath, C.A. Model predictive control for the dynamic encirclement of a target. In Proceedings of the 2012 IEEE American Control Conference (ACC), Montreal, QC, Canada, 27–29 June 2012; pp. 2004–2009.
- 28. Ren, W.; Beard, R.W. Decentralized scheme for spacecraft formation flying via the virtual structure approach. *J. Guid. Contro Dynam.* **2004**, *27*, 73–82. [CrossRef]
- 29. Chen, Y.Q.; Wang, Z. Formation control: A review and a new consideration. In Proceedings of the IEEE/RSJ International Conference on Intelligent Robots and Systems. (IROS 2005), Edmonton, AB, Canada, 2–6 August 2005; pp. 3181–3186.
- Kendall, A.G.; Salvapantula, N.N.; Stol, K.A. On-Board Object Tracking Control of a Quadcopter with Monocular Vision. In Proceedings of the International Conference on Unmanned Aircraft Systems (ICUAS), Orlando, FL, USA, 27–30 May 2014.
- Zhang, M.; Liu, H.H.T. Vision-Based Tracking and Estimation of Ground Moving Target Using Unmanned Aerial Vehicle. In Proceedings of the American Control Conference, Baltimore, MD, USA, 30 June–2 July 2010.
- 32. Wang, X.; Zhu, H.; Zhang, D.; Zhou, D.; Wang, X. Vision-based Detection and Tracking of a Mobile Ground Target Using a Fixed-wing UAV. *Int. J. Adv. Robot. Syst.* **2014**, *11*, 156. [CrossRef]
- Hausman, K. Cooperative Occlusion-Aware Multi-Robot Target Tracking using Optimization. In Proceedings of the IEEE/RSJ International Conference on Intelligent Robots and Systems, Hamburg, Germany, 28 September–2 October 2015.
- 34. Ahmed, M.; Subbarao, K. Target Tracking in 3D Using Estimation Based Nonlinear Control Laws for UAVs. *Aerospace* **2016**, *3*, 5. [CrossRef]
- 35. Dobrokhodov, V.N.; Kaminer, I.I.; Jones, K.D. Vision-Based Tracking and Motion Estimation for Moving Targets Using Small UAVs. *J. Guid. Control Dynam.* **2008**, *31*, 907–917. [CrossRef]
- 36. Shtark, T.; Gurfil, P. Tracking a Non-Cooperative Target Using Real-Time Stereovision-Based Control: An Experimental Study. *Sensors* **2017**, *17*, 735. [CrossRef]
- Gurcuoglu, U.; Puerto-Souza, G.A.; Morbidi, F.; Mariottini, G.L. Hierarchical Control of a Team of Quadrotors for Cooperative Active Target Tracking. In Proceedings of the IEEE/RSJ International Conference on Intelligent Robots and Systems, Tokyo, Japan, 3–7 Novomer 2013.
- 38. Al-Isawi, M.M.A.; Sasiadek, J.Z. Guidance and Control of a Robot Capturing an Uncooperative Space Target. *J. Intell. Robot. Syst.* **2019**, *93*, 713–721. [CrossRef]

- Ding, S.; Liu, G.; Li, Y.; Zhang, J.; Yuan, J.; Sun, F. SLAM and Moving Target Tracking Based on Constrained Local Submap Filter. In Proceedings of the 2015 IEEE International Conference on Information and Automation, Lijiang, China, 8–10 August 2015.
- 40. Ahmad, A.; Tipaldi, G.D.; Lima, P.; Burgard, W. Cooperative robot localization and target tracking based on least squares minimization. In Proceedings of the 2013 IEEE International Conference on Robotics and Automation, Karlsruhe, Germany, 6–10 May 2013.
- 41. Gohring, D.; Burkhard, H.D. Multi Robot Object Tracking and Self Localization Using Visual Percept Relations. In Proceedings of the 2006 IEEE/RSJ International Conference on Intelligent Robots and Systems, Beijing, China, 9–15 October 2006.
- 42. Chakraborty, A.; Sharma, R.; Brink, K. Target-Centric Formation Control in GPS-denied Environments. In Proceedings of the 2018 AIAA Information Systems-AIAA Infotech @ Aerospace, Kissimmee, FL, USA, 8–12 January 2018.
- 43. Nielsen, J.; Beard, R. Relative Moving Target Tracking and Circumnavigation. In Proceedings of the 2019 American Control Conference (ACC), Philadelphia, PA, USA, 10–12 July 2019; pp. 1122–1127.
- 44. Pak, J.M. Gaussian Sum FIR Filtering for 2D Target Tracking. *Int. J. Control Autom. Syst.* **2019**, *18*, 643–649. [CrossRef]
- 45. Wang, S.,; Jiang, F.; Zhang, B.; Ma, R.; Hao, Q. Development of UAV-Based Target Tracking and Recognition Systems. *IEEE Trans. Intell. Transp. Syst.* **2019**. [CrossRef]
- Yang, Z.; Zhu, S.; Chen, C.; Guan, X.; Feng, G. Distributed Formation Target Tracking in Local Coordinate Systems. In Proceedings of the 2019 IEEE 15th International Conference on Control and Automation (ICCA), Edinburgh, UK, 16–19 July 2019; pp. 840–845.
- 47. Reif, K.; Günther, S.; Yaz, E.; Unbehauen, R. Stochastic stability of the discrete-time extended Kalman filter. *IEEE Trans. Autom. Control* **1999**, *44*, 714–728. [CrossRef]
- 48. Kluge, S.; Reif, K.; Brokate, M. Stochastic stability of the extended Kalman filter with intermittent observations. *IEEE Trans. Autom. Control* **2010**, *55*, 514–518. [CrossRef]
- 49. Khalil, H.K. Nonlinear Systems, 3rd ed.; Prentice-Hall: Upper Saddle River, NJ, USA, 2002.
- Euston, M.; Coote, P.; Mahony, R.; Kim, J.; Hamel, T. A complementary filter for attitude estimation of a fixed-wing UAV. In Proceedings of the IEEE/RSJ International Conference on Intelligent Robots and Systems, Nice, France, 22–26 September 2008; pp. 340–345. doi:10.1109/iros.2008.4650766. [CrossRef]
- 51. Munguia, R.; Grau, A. A Practical Method for Implementing an Attitude and Heading Reference System. *Int. J. Adv. Robot. Syst.* **2014**, *11*, 62. [CrossRef]
- 52. Hartley, R.; Zisserman, A. *Multiple View Geometry in Computer Vision*, 2nd ed.; Cambridge University Press: Cambridge, MA, USA, 2003.
- 53. Srisamosorn, V.; Kuwahara, N.; Yamashita, A.; Ogata, T. Human-tracking System Using Quadrotors and Multiple Environmental Cameras for Face-tracking Application. *Int. J. Adv. Robot. Syst.* **2017**, *14*, 1729881417727357. [CrossRef]
- 54. Benezeth, Y.; Emile, B.; Laurent, H.; Rosenberger, C. Vision-Based System for Human Detection and Tracking in Indoor Environment. *Int. J. Soc. Robot.* **2010**, *2*, 41–52. [CrossRef]
- Olivares-Mendez, M.A.; Fu, C.; Ludivig, P.; Bissyandé, T.F.; Kannan, S.; Zurad, M.; Annaiyan, A.; Voos, H.; Campoy, P. Towards an Autonomous Vision-Based Unmanned Aerial System against Wildlife Poachers. Sensors 2015, 15, 31362–31391. [CrossRef]
- Briese, C.; Seel, A.; Andert, F. Vision-based detection of non-cooperative UAVs using frame differencing and temporal filter. In Proceedings of the International Conference on Unmanned Aircraft Systems, Dallas, TX, USA, 12–15 June 2018.
- Mejías, L.; McNamara, S.; Lai, J. Vision-based detection and tracking of aerial targets for UAV collision avoidance. In Proceedings of the IEEE/RSJ International Conference on Intelligent Robots and Systems, Taipei, Taiwan, 18–22 October 2010; pp.87–92.
- Stateczny, A. Neural Manoeuvre Detection of the Tracked Target in ARPA Systems. *Elsevier* 2002, 34, 209–214. [CrossRef]
- 59. Emran, B.J.; Yesilderik, A. Robust Nonlinear Composite Adaptive Control of Quadrotor. *Int. J. Digital Inf. Wirel. Commun.* 2014, 4, 213–225. [CrossRef]
- 60. Hermann, R.; Krener, A. Nonlinear controllability and observability. *IEEE Trans. Autom. Control* **1977**, 22, 728–740. [CrossRef]

- 61. Slotine, J.E.; Li, W. *Applied Nonlinear Control*; Prentice-Hall Englewood Cliffs: Upper Saddle River, NJ, USA, 1991.
- 62. Durrant-Whyte, H.; Bailey, T. Simultaneous localization and mapping: Part i. *IEEE Robot. Autom. Mag.* **2006**, *13*, 99–110. [CrossRef]
- 63. Bailey, T.; Durrant-Whyte, H. Simultaneous localization and mapping (slam): Part ii. *IEEE Robot. Autom. Mag.* **2006**, *13*, 108–117. [CrossRef]
- 64. Guilietti, F.; Pollini, L.; Innocenti, M. Autonomous formation flight. IEEE Control Syst. Mag. 2000, 20, 30-44.
- 65. Levant, A. Higher-order sliding modes, differentiation and output-feedback control. *Int. J. Control.* **2003**, *76*, 924–941. [CrossRef]
- 66. Vega, L.; Castillo Toledo, B.; Loukianov, A.G. Robust block second order sliding mode control for a quadrotor. *J. Franklin Inst.* **2012**, 349, 719–739. [CrossRef]
- 67. Montiel, J.M.M.; Civera, J.; Davison, A. Unified inverse depth parametrization for monocular SLAM. In Proceedings of the Robotics: Science and Systems Conference, Ann Arbor, MI, USA, 18–22 June 2006.
- Trujillo, J.C.; Munguia, R.; Ruiz-Velázquez, E.; Castillo-Toledo, B. A Cooperative Aerial Robotic Approach for Tracking and Estimating the 3D Position of a Moving Object by Using Pseudo-Stereo Vision. *J. Intell. Robot. Syst.* 2019, *96*, 297–313. [CrossRef]
- 69. Lanzisera, S.; Zats, D.; Pister, K.S.J. Radio frequency time-of-flight distance measurement for low-cost wireless sensor localization. *IEEE Sensors J.* **2011**, *11*, 837–845. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).